

# DECISION

*Fair Work Act 2009* s.185—Enterprise agreement

# Australian Rail Track Corporation Ltd T/A Australian Rail Track Corporation Ltd (AG2017/6055)

# ARTC SOUTH AUSTRALIA / WESTERN AUSTRALIA INFRASTRUCTURE MAINTENANCE ENTERPRISE AGREEMENT 2017

Rail industry

COMMISSIONER HARPER-GREENWELL

MELBOURNE, 23 APRIL 2018

Application for approval of the ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017.

[1] An application has been made for approval of an enterprise agreement known as the *ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017* (the Agreement). The application was made pursuant to s.185 of the *Fair Work Act 2009* (the Act). It has been made by Australian Rail Track Corporation Ltd T/A Australian Rail Track Corporation Ltd. The Agreement is a single enterprise agreement.

[2] The Applicant has provided written undertakings. A copy of the undertakings is attached in Annexure A. I am satisfied that the undertakings will not cause financial detriment to any employee covered by the Agreement and that the undertakings will not result in substantial changes to the Agreement.

[3] Subject to the undertakings referred to above, I am satisfied that each of the requirements of ss.186, 187, 188 and 190 as are relevant to this application for approval have been met. The Agreement does not cover all of the employees of the employer, however, taking into account the factors in Section 186(3) and (3A) I am satisfied that the group of employees was fairly chosen.

[4] The Australian Rail, Tram and Bus Industry Union & "Automotive, Food, Metals, Engineering, Printing and Kindred Industries Union" known as the Australian Manufacturing Workers' Union (AMWU) being a bargaining representatives for the Agreement, have given notice under s.183 of the Act that they want the Agreement to cover them. In accordance with s.201(2) I note that the Agreement covers the organisations.

[5] The Agreement was approved on 23 April 2018 and, in accordance with s.54, will operate from 30 April 2018. The nominal expiry date of the Agreement is 30 April 2021.



**COMMISSIONER** 

Printed by authority of the Commonwealth Government Printer

<AE428104 PR602201>

**Annexure** A

IN THE FAIR WORK COMMISSION

FWC Matter No.: AG2017/6055

Applicant:

Australian Rail Track Corporation

# **Undertaking-section 190**

I, Nathan Farr, General Manager People, of Australian Rail Track Corporation give the following undertakings with respect to the ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 ("the Agreement"):

- I have the authority given to me by Australian Rail Track Corporation to provide this undertaking in relation to this application before the Fair Work Commission.
- In respect of Employees covered by the Agreement, Australian Rail Track Corporation undertakes that:
  - a) Notwithstanding clause 2.7, employees with at least 2 years' but less than 3 years' service are entitled to redundancy in accordance with the National Employment Standards.
  - b) Notwithstanding clause 3.1.2.3, Adult Apprentices will be paid at the rate in the Rail Industry Award if that rate is higher than the rate in clause 3.1.2.3.
  - c) A shiftworker who is a seven day shiftworker, regularly rostered to work on Sundays and public holidays, will receive an additional weeks' annual leave.
  - d) Notwithstanding Clause 4.1.3, an employee can accrue up to 8 weeks (10 weeks for shiftworkers) annual leave before an employee has an excessive leave accrual which can result in a direction to take leave with notice of not less than 8 weeks or more than 12 months.
  - e) Where an employee's employment is terminated in accordance with clause 4.2.5.2 the employee will receive the applicable notice or payment in lieu.
  - f) Notwithstanding clause 4.5.2, the taking of Community Service Leave in relation to an emergency service is not dependent on whether the employee is required to attend to Australian Rail Track Corporation assets.

1

Employee name: Nathan Farr

Signature: Date: 10 April 2018

Note - this agreement is to be read together with an undertaking given by the employer. The undertaking is taken to be a term of the agreement. A copy of it can be found at the end of the agreement.

Australian Rail Track Corporation

# ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page 1 of 53

# Contents

1. INTRODUCTION	7
1.1 INTRODUCTION	7
1.1.1 What will this Agreement be officially known as?	7
1.1.2 Who is covered by this Agreement?	7
1.1.3 Who is bound by this Agreement?	7
1.1.4 What do the words mean in this Agreement?	7
1.2 DURATION OF AGREEMENT	8
1.2.1 What is the duration of this Agreement?	8
1.3 OBJECTIVES AND OBLIGATIONS UNDER THIS AGREEMENT	8
1.3.1 What are the main objectives of this Agreement?	8
1.3.2 What are my obligations?	9
1.3.3 What is the Commitment to Improving Productivity, Efficiency and Flexibility?	9
2. HOW WE WILL WORK	10
2.1 PAYMENT OF SALARY	. 10
2.1.1 When will I be paid?	10
2.1.2 How will I be paid?	10
2.2 EMPLOYMENT CATEGORIES	. 10
2.2.1 What categories of employment could I be employed under in this Agreement?	10
2.2.1.1 What is a 'full-time' Employee?	10
2.2.1.2 What is a 'part-time' Employee?	10
2.2.1.3 What is a 'temporary' Employee?	11
2.3 WHAT PROVISIONS RELATE TO CASUAL EMPLOYMENT?	. 11
2.3.1 What will I be paid if I am a casual employee?	11
2.4 WHAT PROBATIONARY PERIOD IS APPLICABLE TO MY EMPLOYMENT?	. 11
2.4.1 What happens if ARTC is dissatisfied with my employment during my probationary	
period?	
2.5 NEW EMPLOYEES	
2.5.1 What are my duties as an employee at ARTC?	
2.5.2 What is Abandonment of Employment?	
2.5.3 What if I am unable to notify ARTC of an absence due to unforeseen circumstances?	
2.6 TERMINATION OF EMPLOYMENT	
2.6.1 What if my employment is to be terminated?	
2.6.2 What notice periods apply in relation to the termination of my employment?	
2.6.2.1 What is the required period of notice?	
2.6.2.2 Will the period of notice be increased if I am over 45 years old and have completed a least 2 years continuous service?	
2.6.3 How is the period of continuous service ascertained?	13

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page 2 of 53

.

2.6.4 Can I receive payment in lieu of notice?	13
2.6.5 How is payment in lieu of notice calculated?	13
2.7 REDUNDANCY PAY	13
2.7.1 What entitlement will I receive if my position is made redundant?	13
2.7.2 What other entitlements apply to me in a redundancy situation?	14
2.8 APPRENTICES	14
2.8.1 What provisions apply to me as an Apprentice?	14
2.8.2 What apprenticeship trades are applicable under this Agreement?	14
2.9 HOURS OF WORK	15
2.9.1 What are my ordinary hours of work over a 28 day period?	15
2.9.2 Am I entitled to rostered days off?	15
2.9.3 What if I am required to work on my RDO?	15
2.9.4 Can I accumulate my RDOs and add them to my annual leave?	15
2.10 ROSTERING	15
2.10.1 Can ARTC roster me to work afternoons, nights, weekends and public holidays?	15
2.10.2 Can I be required to work at places other than my designated work location?	16
2.10.3 Will I be consulted about my roster?	16
2.11 MEAL BREAKS	17
2.11.1 What meal breaks am I entitled to?	
2.11.2 What other rest breaks am I entitled to?	17
2.12 NOTICE	17
2.12.1 What period of notice applies where I am required to work at a location other than	17 my
	<i>17</i> my 17
2.12.1 What period of notice applies where I am required to work at a location other than designated work location?	17 my 17 17
<ul><li>2.12.1 What period of notice applies where I am required to work at a location other than designated work location?</li><li>2.13 ALLOWANCES.</li></ul>	17 my 17 17 17
<ul> <li>2.12.1 What period of notice applies where I am required to work at a location other than designated work location?</li></ul>	17 my 17 17 17 17
<ul> <li>2.12.1 What period of notice applies where I am required to work at a location other than designated work location?</li> <li>2.13 ALLOWANCES.</li> <li>2.13.1 On-Call / Call Out</li></ul>	17 my 17 17 17 17 17
<ul> <li>2.12.1 What period of notice applies where I am required to work at a location other than designated work location?</li> <li>2.13 ALLOWANCES.</li> <li>2.13.1 On-Call / Call Out</li></ul>	17 my 17 17 17 17 17 18
<ul> <li>2.12.1 What period of notice applies where I am required to work at a location other than designated work location?</li></ul>	17 my 17 17 17 17 17 18 18
<ul> <li>2.12.1 What period of notice applies where I am required to work at a location other than designated work location?</li></ul>	17 my 17 17 17 17 17 18 18
<ul> <li>2.12.1 What period of notice applies where I am required to work at a location other than designated work location?</li></ul>	17 my 17 17 17 17 18 18 18
<ul> <li>2.12.1 What period of notice applies where I am required to work at a location other than designated work location?</li> <li>2.13 ALLOWANCES.</li> <li>2.13.1 On-Call / Call Out</li></ul>	17 my 17 17 17 17 18 18 18 18 18
<ul> <li>2.12.1 What period of notice applies where I am required to work at a location other than designated work location?</li> <li>2.13 ALLOWANCES.</li> <li>2.13.1 On-Call / Call Out</li></ul>	17 my 17 17 17 17 17 18 18 18 18 18 19 19
<ul> <li>2.12.1 What period of notice applies where I am required to work at a location other than designated work location?</li></ul>	17 my 17 17 17 17 17 18 18 18 18 18 19 19 19
<ul> <li>2.12.1 What period of notice applies where I am required to work at a location other than designated work location?</li> <li>2.13 ALLOWANCES.</li> <li>2.13.1 On-Call / Call Out</li> <li>2.13.1.1 Will I be required to participate in on-call rosters and attend call outs?</li> <li>2.1413.1.2 What will I be paid if I am on-call?</li> <li>2.13.2 What will I be paid for a call-out?</li> <li>2.13.2 What applies to me if I am required to work on a Saturday?</li> <li>2.13.4 What about a rest break whilst working overtime?</li> <li>2.14 SHIFT WORK.</li> <li>2.14.1 What is shift work?</li> </ul>	17 my 17 17 17 17 17 18 18 18 18 19 19 19 19
<ul> <li>2.12.1 What period of notice applies where I am required to work at a location other than i designated work location?</li></ul>	17 my 17 17 17 17 17 18 18 18 18 19 19 19 19 19 19
<ul> <li>2.12.1 What period of notice applies where I am required to work at a location other than designated work location?</li></ul>	17 my 17 17 17 17 17 17 18 18 18 18 18 19 19 19 19 19 19

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **3** of **53** 

3.	REMUNERATION AND RELATED	.21
	3.1 CLASSIFICATION AND SALARY STRUCTURE	. 21
	3.1.1 What is the salary structure at ARTC?	.21
	3.1.1.1 When are salary increases applicable?	. 21
	3.1.2 What salary increases will I receive over the term of this Agreement?	21
	3.1.2.1 Non-Supervisory Staff:	21
	3.1.2.2 Supervisory Staff:	21
	3.1.2.3 I'm an Apprentice, what is my salary?	22
	3.1.3 What is included in my Total Remuneration Package?	22
	3.1.4 What is the Competency Structure for Non-Supervisory Staff at ARTC?	22
	3.1.4.1 How do I apply for competency assessment?	23
	3.1.4.2 What if I don't have my log book (where required) and/or training records when I am due to be assessed?	
	3.2 OVERTIME	. 23
	3.2.1 Will I be required to work overtime?	23
	3.2.2 What compensation will I receive for working overtime?	23
	3.2.3 What is the rest period after working overtime?	24
	3.2.4 What if I'm required to work without having had 10 consecutive hours off?	24
	3.2.5 What transport arrangements apply after working overtime?	24
	3.3 TRAVELLING AWAY FROM HOME	. 24
	3.3.1 Can I be required to work at locations which will temporarily require me to travel away from home and stay overnight?	
	3.3.1.1 What happens if I am required to travel away from home and stay overnight?	25
	3.3.1.2 What will I be paid when I am required to travel away from home and stay overnight	? 25
	3.4 CARRYING OUT HIGHER DUTIES	. 26
	3.4.1 How will I be paid if I am carrying out higher duties?	26
	4 BALANCING WORK AND LIFE	. 26
	4.1 ANNUAL LEAVE	. 26
	4.1.1 What is my entitlement to Annual Leave?	26
	4.1.1.1 I work shift work and weekends, do I get additional annual leave?	26
	4.1.2 Under what methods can I take Annual Leave?	26
	4.1.3 How much leave can I accrue?	27
	4.1.4 Can I cash out my annual leave without taking time off?	27
	4.1.5 What proportionate leave applies on termination?	27
	4.1.6 What happens if there is a break in my service?	27
	4.1.7 How is continuous service calculated?	27
	4.1.8 What will I be paid whilst on leave?	28
	4.1.9 Is there an Annual Close Down?	28

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page 4 of 53

	4.1.10 When does leave commence when I'm on a distant job?	28
	4.1.11 Do I need to take leave?	28
	4.1.12 What if I am sick whilst on annual leave?	. 29
	4.1.13 How do I notify of my annual leave preferences in advance?	. 29
	4.1.14 What if I don't post my annual leave preferences in advance?	. 29
4	4.2 PERSONAL LEAVE INCORPORATING SICK LEAVE AND CARER LEAVE	. 29
	4.2.1 What is my personal/carer leave entitlement?	. 29
	4.2.2 What is carer leave?	. 29
	4.2.2.1 What does "immediate family or household member" mean?	. 29
	4.2.3 What is personal leave?	. 30
	4.2.4 How is personal/carer leave used?	.30
	4.2.5 What about sick leave?	30
	4.2.5.1 What do I do if I am sick or need to claim carer leave?	30
	4.2.5.2 What happens if I am sick for a long period of time?	30
4	4.3 COMPASSIONATE AND BEREAVEMENT LEAVE	. 30
	4.3.1 What are compassionate and bereavement leave and to what extent am I covered?	30
	4.3.2 What if I need leave to cover bereavement or compassionate circumstances for someo close to me but they are not an immediate family or household member?	
2	1.4 PARENTAL LEAVE	. 31
4	4.5 OTHER LEAVE	. 31
	4.5.1 What leave am I entitled to if I am required to serve on a jury (jury duty)?	31
	4.5.2 Am I entitled to emergency services leave?	.31
	4.5.3 Can I take military leave?	32
4	4.6 PUBLIC HOLIDAYS	. 32
	4.6.1 What is my entitlement to public holidays?	3Ź
	4.6.2 What if the public holidays fall on a weekend?	. 32
	4.6.3 Can public holidays be substituted?	.32
	4.6.4 What will I be paid if I am required to work on a public holiday?	32
	ARE CASUAL EMPLOYEES ENTITLED TO ANY PAID HOLIDAYS UNDER THIS	33
4	1.8 LONG SERVICE LEAVE	. 33
	4.8.1 What is my entitlement to Long Service Leave?	.33
	4.8.2 Can I cash in Long Service Leave?	34
5. I	HEALTH WELLBEING AND ASSESSMENT	34
ł	5.1 HEALTH ASSESSMENT	. 34
	5.1.1 When do I have to have a medical?	34
	5.1.2 Do I have to pay for these Health Assessments?	34
	5.1.3 When do I have to have my health assessment?	34

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page 5 of 53

5.1.4 What happens to me if I am deemed unfit for work?	34
5.2 WORK CLOTHING AND PERSONAL PROTECTIVE EQUIPMENT (PPE)	34
5.2.1Will I be given clothes and PPE for work?	34
5.2.2 Who is responsible for my issued work clothes and PPE?	34
5.2.3 What do I do if I need replacements for worn, defective or damaged work and safety gear?	35
5.2.4 What clothes and safety gear will I be given?	35
5.2.5 Who is responsible for washing my PPE and work clothes?	35
5.2.6 Do I get to keep the work clothes and safety gear?	35
5.3 INCLEMENT WEATHER	35
5.3.1 What does "Inclement Weather" Mean?	35
5.3.2 How is an inclement weather situation managed?	35
6. RECOGNISING YOUR CONTRIBUTION	36
6.1 TRAINING AND DEVELOPMENT	36
6.1.1 Will ARTC provide opportunities for learning and career development?	36
6.1.2 How do I get access to training opportunities?	37
6.2 BONUSES	37
6.2.1 Can I participate in ARTC's Bonus Plan?	37
7. CONSULTATION AND DISPUTE RESOLUTION	37
7.1 WILL I BE CONSULTED ON ANY MAJOR CHANGES HAPPENING AT ARTC?	37
THE WEET BE CONSISTED ON ANY MAJOR ON MOLETING THE MACHINE THE	
7.2 CONSULTATION PROCESS	
	39
7.2 CONSULTATION PROCESS	<b>39</b> 39
7.2 CONSULTATION PROCESS	39 39 40
7.2 CONSULTATION PROCESS	<b>39</b> 39 <i>40</i> 40
7.2 CONSULTATION PROCESS       3         7.2.1 What is ARTC's Consultation Process       3         7.3 CONSULTATIVE COMMITTEE       4         7.3.1 What is the purpose of the Consultative Committee?       4	39 39 40 40 40
7.2 CONSULTATION PROCESS       3         7.2.1 What is ARTC's Consultation Process       3         7.3 CONSULTATIVE COMMITTEE       4         7.3.1 What is the purpose of the Consultative Committee?       4         7.3.2 What is the composition of the Consultative Committee?       4	39 39 40 40 40 40
7.2 CONSULTATION PROCESS       3         7.2.1 What is ARTC's Consultation Process       3         7.3 CONSULTATIVE COMMITTEE       4         7.3.1 What is the purpose of the Consultative Committee?       4         7.3.2 What is the composition of the Consultative Committee?       4         7.3.3 What are the scope of tasks undertaken by the Consultative Committee?       4	39 39 40 40 40 40 40
7.2 CONSULTATION PROCESS       3         7.2.1 What is ARTC's Consultation Process.       3         7.3 CONSULTATIVE COMMITTEE       4         7.3.1 What is the purpose of the Consultative Committee?       4         7.3.2 What is the composition of the Consultative Committee?       4         7.3.3 What are the scope of tasks undertaken by the Consultative Committee?       4         7.4 TOOL BOX MEETINGS       4	<ul> <li>39</li> <li>39</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> </ul>
7.2 CONSULTATION PROCESS       3         7.2.1 What is ARTC's Consultation Process       3         7.3 CONSULTATIVE COMMITTEE       4         7.3.1 What is the purpose of the Consultative Committee?       4         7.3.2 What is the composition of the Consultative Committee?       4         7.3.3 What are the scope of tasks undertaken by the Consultative Committee?       4         7.4 TOOL BOX MEETINGS       4         7.4.1 What are Tool Box Meetings and what is their purpose?       4	<ul> <li>39</li> <li>39</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>41</li> </ul>
7.2 CONSULTATION PROCESS       3         7.2.1 What is ARTC's Consultation Process.       3         7.3 CONSULTATIVE COMMITTEE       4         7.3.1 What is the purpose of the Consultative Committee?       4         7.3.2 What is the composition of the Consultative Committee?       4         7.3.3 What are the scope of tasks undertaken by the Consultative Committee?       4         7.4 TOOL BOX MEETINGS       4         7.4.1 What are Tool Box Meetings and what is their purpose?       4         7.5 DISPUTE SETTLEMENT PROCEDURE       4	<ul> <li>39</li> <li>39</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>41</li> </ul>
7.2 CONSULTATION PROCESS       3         7.2.1 What is ARTC's Consultation Process.       3         7.3 CONSULTATIVE COMMITTEE       4         7.3.1 What is the purpose of the Consultative Committee?       4         7.3.2 What is the composition of the Consultative Committee?       4         7.3.3 What are the scope of tasks undertaken by the Consultative Committee?       4         7.4 TOOL BOX MEETINGS.       4         7.5 DISPUTE SETTLEMENT PROCEDURE       4         7.5.1 What process do I follow to resolve a dispute in relation to my work with ARTC?       4	<ul> <li>39</li> <li>39</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>41</li> <li>41</li> <li>42</li> </ul>
7.2 CONSULTATION PROCESS       3         7.2.1 What is ARTC's Consultation Process       3         7.3 CONSULTATIVE COMMITTEE       4         7.3.1 What is the purpose of the Consultative Committee?       4         7.3.2 What is the composition of the Consultative Committee?       4         7.3.3 What are the scope of tasks undertaken by the Consultative Committee?       4         7.4 TOOL BOX MEETINGS       4         7.5 DISPUTE SETTLEMENT PROCEDURE       4         7.5.1 What process do I follow to resolve a dispute in relation to my work with ARTC?       4         8. FLEXIBILITY TERM       4	<ul> <li>39</li> <li>39</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>41</li> <li>42</li> <li>42</li> </ul>
7.2 CONSULTATION PROCESS       3         7.2.1 What is ARTC's Consultation Process.       3         7.3 CONSULTATIVE COMMITTEE       4         7.3.1 What is the purpose of the Consultative Committee?       4         7.3.2 What is the composition of the Consultative Committee?       4         7.3.3 What are the scope of tasks undertaken by the Consultative Committee?       4         7.4 TOOL BOX MEETINGS.       4         7.5 DISPUTE SETTLEMENT PROCEDURE       4         7.5.1 What process do I follow to resolve a dispute in relation to my work with ARTC?       4         8. FLEXIBILITY TERM       4         8.1 INDIVIDUAL FLEXIBILITY ARRANGEMENT.       4	<ul> <li>39</li> <li>39</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>41</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> </ul>
7.2 CONSULTATION PROCESS       3         7.2.1 What is ARTC's Consultation Process       3         7.3 CONSULTATIVE COMMITTEE       4         7.3.1 What is the purpose of the Consultative Committee?       4         7.3.2 What is the composition of the Consultative Committee?       4         7.3.3 What are the scope of tasks undertaken by the Consultative Committee?       4         7.4 TOOL BOX MEETINGS       4         7.5 DISPUTE SETTLEMENT PROCEDURE       4         7.5.1 What process do I follow to resolve a dispute in relation to my work with ARTC?       4         8. FLEXIBILITY TERM       4         8.1 INDIVIDUAL FLEXIBILITY ARRANGEMENT       4         8.1.1 Can the terms of this Agreement be varied?       4	<ul> <li>39</li> <li>39</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>41</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> </ul>
7.2 CONSULTATION PROCESS       3         7.2.1 What is ARTC's Consultation Process       3         7.3 CONSULTATIVE COMMITTEE       4         7.3.1 What is the purpose of the Consultative Committee?       4         7.3.2 What is the composition of the Consultative Committee?       4         7.3.3 What are the scope of tasks undertaken by the Consultative Committee?       4         7.4 TOOL BOX MEETINGS       4         7.4.1 What are Tool Box Meetings and what is their purpose?       4         7.5 DISPUTE SETTLEMENT PROCEDURE       4         7.5.1 What process do I follow to resolve a dispute in relation to my work with ARTC?       4         8. FLEXIBILITY TERM       4         8.1 INDIVIDUAL FLEXIBILITY ARRANGEMENT       4         8.1.2 What is an Individual Flexibility Arrangement?       4	<ul> <li>39</li> <li>39</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>41</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> </ul>
7.2 CONSULTATION PROCESS       2         7.2.1 What is ARTC's Consultation Process       2         7.3 CONSULTATIVE COMMITTEE       2         7.3.1 What is the purpose of the Consultative Committee?       2         7.3.2 What is the composition of the Consultative Committee?       2         7.3.3 What are the scope of tasks undertaken by the Consultative Committee?       2         7.4 TOOL BOX MEETINGS       2         7.4.1 What are Tool Box Meetings and what is their purpose?       2         7.5 DISPUTE SETTLEMENT PROCEDURE       2         7.5.1 What process do I follow to resolve a dispute in relation to my work with ARTC?       2         8. FLEXIBILITY TERM       2         8.1 INDIVIDUAL FLEXIBILITY ARRANGEMENT       2         8.1.2 What is an Individual Flexibility Arrangement?       2         8.1.3 How do I make an Individual Flexibility Arrangement?       2	<ul> <li>39</li> <li>39</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>41</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> <li>43</li> </ul>
7.2 CONSULTATION PROCESS       2         7.2.1 What is ARTC's Consultation Process.       2         7.3 CONSULTATIVE COMMITTEE       2         7.3.1 What is the purpose of the Consultative Committee?       2         7.3.2 What is the composition of the Consultative Committee?       2         7.3.3 What are the scope of tasks undertaken by the Consultative Committee?       2         7.4 TOOL BOX MEETINGS       2         7.4.1 What are Tool Box Meetings and what is their purpose?       2         7.5 DISPUTE SETTLEMENT PROCEDURE       2         7.5.1 What process do I follow to resolve a dispute in relation to my work with ARTC?       2         8. FLEXIBILITY TERM       2         8.1 INDIVIDUAL FLEXIBILITY ARRANGEMENT       2         8.1.2 What is an Individual Flexibility Arrangement?       2         8.1.4 How do I make an Individual Flexibility Arrangement?       2         8.1.4 How do I terminate (end) an Individual Flexibility Arrangement?       2	<ul> <li>39</li> <li>39</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>40</li> <li>41</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> <li>42</li> <li>43</li> </ul>

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page 6 of 53

# **1. INTRODUCTION**

# **1.1 INTRODUCTION**

# 1.1.1 What will this Agreement be officially known as?

This Agreement will be known as the ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 ("Agreement").

# 1.1.2 Who is covered by this Agreement?

You are covered by this Agreement if you are an Infrastructure Maintenance employee of Australian Rail Track Corporation (ARTC) employed in one of the classifications listed in clauses 3.1.2.1, 3.1.2.2 or 3.1.2.3 of this Agreement and your designated work location is within South Australia or Western Australia.

### 1.1.3 Who is bound by this Agreement?

This Agreement will be binding upon ARTC, and employees engaged in any of the classifications/occupations/levels referred to in clause 3.1.1.

# 1.1.4 What do the words mean in this Agreement?

### The definitions of the words used in this Agreement are as follows:

'Act' means the Fair Work Act 2009 (Commonwealth).

'ARTC' means Australian Rail Track Corporation.

*'Designated Work Location'* means the provisioning centre or other location you are appointed to and where you are ordinarily expected to report for duty.

*'Employee'* means an employee of Australian Rail Track Corporation whose designated work location is within South Australia or Western Australia and who satisfies clause 1.1.2 of this Agreement.

*National Employment Standards'* means the minimum employment standards as contained in the *Fair Work Act 2009.* 

*Nominal Salary*' means an employee's Total Annual Remuneration Package (TRP), less the applicable superannuation guarantee contribution component.

'Roster Cycle' means the designated period over which work is arranged.

'Rostered Employee' means an employee who performs 'Rostered Work'.

*Rostered Day Off* is a non-working day derived through sufficient accumulation of ordinary work time as per clause 2.9.1 of this Agreement.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page 7 of 53 *Rostered Work* means work performed on a pre-planned rostered basis, which includes ordinary working time involving shifts and/or weekends and public holidays.

*Suitable Evidence'* means a medical certificate or statutory declaration to substantiate evidence of a personal illness or injury.

'*TRP*' means Total annual Remuneration Package, as described in clause 3.1.3

*'This Agreement'* means ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017.

### **1.2 DURATION OF AGREEMENT**

### **1.2.1 What is the duration of this Agreement?**

This Agreement will come into effect 7 days after the date of approval by the Fair Work Commission and its nominal expiry date will be 3 years from that date.

# **1.3 OBJECTIVES AND OBLIGATIONS UNDER THIS AGREEMENT**

#### **1.3.1 What are the main objectives of this Agreement?**

The main objectives of this Agreement are:

- To promote the growth, efficiency and effectiveness of the rail industry as a competitive, value-added segment of the national land transport logistics network;
- To promote an efficient and safe working environment;
- To enhance the value of ARTC through being a competitive, flexible and innovative company;
- To provide a workplace that promotes real gains in productivity, efficiency and flexibility; and
- To establish a workplace which enables variety, skills and job security for employees by:
  - Developing and maintaining a workplace which encourages and facilitates teamwork, personal and job skill enhancement to achieve the company's objectives;
  - Promoting efficient and effective delivery of services to ARTC's customers;
  - Implementing change through constructive consultation to ensure a profitable, competitive and efficient operation;
  - Providing employees with competitive marketplace-based remuneration.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page 8 of 53

- Enhancing the skill base of the workforce including, where appropriate, through the implementation of a nationally recognised competency based training and career progression system; and
- Achieving continuous improvement in the operations and service delivery of ARTC.

It is also recognised that it is ARTC's right to direct an employee to carry out such duties as are within the scope of the employee's skill, competence and training.

# 1.3.2 What are my obligations?

You will:

- Act as an integral and professional member of the entire team by effectively discharging your responsibilities;
- Be accountable and responsible for your decisions and actions;
- Deliver long-term customer satisfaction by utilising skills endorsed or nominated by ARTC;
- Maintain a high level of proficiency in your area of expertise;
- Commit to implementing ARTC's critical focus on productivity;
- Fully participate in relevant joint problem solving exercises at the workplace;
- Act ethically, constructively and co-operatively with employees, customers and business associates of Australian Rail Track Corporation; and
- Work safely, in consideration of the health and safety of yourself and others.

# 1.3.3 What is the Commitment to Improving Productivity, Efficiency and Flexibility?

Through the ongoing exercise of work group restructuring, the parties to this Agreement are committed to the need to continually develop further flexibility over the functions performed at the ARTC facilities, consistent with agreed productivity objectives and the removal of artificial demarcation work barriers by agreement, subject to competence, training and classification.

The commitment extends to the individual employees performing tasks which, while primarily involving the skill of the employee's classification, are incidental or peripheral to the primary task and enables the completion of the whole task. Such incidental or peripheral tasks would be carried out giving due consideration to it being safe, legal, sensible and within the employee's competence. Furthermore, any such commitment to provide productivity, efficiency and flexibility will not promote deskilling of an employee.

Arrangements made to accommodate flexible working practices arising from the provisions of the Agreement between ARTC and employees will be recorded in Individual Flexibility Arrangements as per clause 8 of this Agreement

# 2. HOW WE WILL WORK

# 2.1 PAYMENT OF SALARY

### 2.1.1 When will I be paid?

You will be paid fortnightly in arrears.

### 2.1.2 How will I be paid?

ARTC will electronically pay you directly into your bank account. When commencing employment, you are required to nominate an account number to enable us to make payment to you into your bank account.

# 2.2 EMPLOYMENT CATEGORIES

# 2.2.1 What categories of employment could I be employed under in this Agreement?

Under this Agreement, you will be employed under one of the following categories:

- Permanent full-time or part-time;
- Temporary full-time or part-time;
  - Fixed-term; or
  - o Specified project.
- Casual

# 2.2.1.1 What is a 'full-time' Employee?

If you are a full-time employee, your ordinary hours of work will be 152 hours over a 28 day period, unless otherwise specified in this Agreement.

### 2.2.1.2 What is a 'part-time' Employee?

If you are a part-time employee, you will be required to work a regular pattern of hours averaging less than 38 hours per week, on a permanent, temporary or fixedterm basis as provided for in this Agreement.

The agreed hours will be the contract hours of work. Contract hours can only be varied in writing, by agreement between you and ARTC.

The daily starting and finishing times and daily/weekly hours of work will be agreed at your commencement of employment and provided to you in writing.

If you are a part-time employee, you will not be required to work additional hours beyond your agreed minimum hours, but you may be offered the opportunity to do so. Where you agree to work additional hours, payment for these hours will be at single time rates up to the number of ordinary hours for an equivalent permanent fulltime employee.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **10** of **53**  Leave entitlements and remuneration will apply pro-rata to part-time employees on the basis that ordinary hours for full-time employees are 38 hours per week.

# 2.2.1.3 What is a 'temporary' Employee?

If you are a temporary employee, you may be employed on a full-time or part-time basis:

- (a) for a fixed period no longer than 24 months; or
- (b) for a specified project, with a specified start and anticipated finish date, in a role not required on an ongoing basis.

An offer of temporary employment will specify the fixed period of the employment or the parameters and expected duration of the project.

Temporary employees will have any untaken annual leave paid out on termination.

# 2.3 WHAT PROVISIONS RELATE TO CASUAL EMPLOYMENT?

A casual employee is to be one engaged and paid as such. Workers employed on a casual basis are engaged on an hourly contract of service.

# 2.3.1 What will I be paid if I am a casual employee?

A casual employee for working ordinary time will be paid per hour 1/38 of the prescribed actual weekly rate for the work performed, plus 25%. A casual employee is not entitled to any paid leave benefits under this Agreement including annual holidays, sick leave, bereavement leave, public or additional holidays, or any redundancy or severance payments.

# 2.4 WHAT PROBATIONARY PERIOD IS APPLICABLE TO MY EMPLOYMENT?

The probation period for new employees is as defined under the *Act* and is six months.

Furthermore, you will be advised of the standards of performance and conduct expected.

During the probationary period, you will be provided with feedback concerning your employment, which will be given by your supervisor/local manager.

# 2.4.1 What happens if ARTC is dissatisfied with my employment during my probationary period?

Where ARTC is dissatisfied with your employment, you will be given the opportunity to improve your performance, conduct or behaviour. If ARTC is dissatisfied with your employment following feedback and reasonable opportunities to improve performance, conduct or behaviour, your employment will cease at the end of the probation period, unless it is terminated earlier in accordance with the appropriate clauses of this Agreement. The reason/s for termination will be recorded.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **11** of **53**  During the review of the unsatisfactory performance, conduct or behaviour, you may request to be represented by an employee representative.

# 2.5 NEW EMPLOYEES

# 2.5.1 What are my duties as an employee at ARTC?

You will perform such work as ARTC will, from time to time, reasonably require, provided you have been trained to the level of skill required to perform such duties and have been evaluated to have the appropriate skills. Where training is required to enable you to undertake any varied duties, you will undertake such training within the classification structure of this Agreement.

### 2.5.2 What is Abandonment of Employment?

The absence of any employee from work for a continuous period exceeding five (5) working days without the consent of ARTC and/or without notification to ARTC, will be prima facie evidence that the employee has abandoned his/her employment. ARTC will maintain a record of attempts made to contact the employee in these circumstances.

# 2.5.3 What if I am unable to notify ARTC of an absence due to unforeseen circumstances?

In the event you are unable to notify ARTC of your absence due to unforseen hospitalisation or similar causes, you will be given the opportunity to explain the reason for a specialised absence which may lead to reinstatement.

# 2.6 TERMINATION OF EMPLOYMENT

### 2.6.1 What if my employment is to be terminated?

Except as provided in clause 2.6.2 of this Agreement, an employee who has been continuously employed for four weeks or longer will not, without the approval of ARTC, leave the service of ARTC until the expiration of one weeks' notice of the employee's intention to do so.

# 2.6.2 What notice periods apply in relation to the termination of my employment?

In order to terminate the employment of an employee, ARTC must comply with either of the following requirements:

- Provide the employee the period of notice set out in 2.6.2.1 and 2.6.2.2 or compensation instead of notice, or;
- Terminate the employee due to serious misconduct, that is, misconduct of a kind such that it would be unreasonable to require ARTC to continue the employment during the notice period.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **12** of **53** 

# 2.6.2.1 What is the required period of notice?

The required period of notice is first calculated using this table:

Period of Continuous Service	Period of Notice
Not more than 1 year	1 week
More than 1 year but not more than 3 years	2 weeks
More than 3 years but not more than 5 years	3 weeks
More than 5 years	4 weeks

# 2.6.2.2 Will the period of notice be increased if I am over 45 years old and have completed at least 2 years continuous service?

Yes. The period of notice is increased by one week if you are over 45 years old and have completed at least 2 years of continuous service with ARTC.

# 2.6.3 How is the period of continuous service ascertained?

Continuous service will be calculated under clause 4.1.7 for the purpose of clause 2.6.2.

# 2.6.4 Can I receive payment in lieu of notice?

Yes. The amount of payment in lieu of notice will not be less than the total of the amounts that, if your employment had continued until the end of the required period of notice, ARTC would have become liable to pay to you because of the employment continuing during that period: provided that where both parties agree to a lesser period of notice, no payment will be due for the agreed reduction of notice.

# 2.6.5 How is payment in lieu of notice calculated?

The total payment in lieu of notice must be worked out on the basis of:

- the employee's ordinary hours of work (even if they are not standard hours); and
- the amounts payable to the employees in respect of those hours, including, where applicable, allowances, and penalties; and
- any other amounts specified in this Agreement and payable under the employees' contract of employment.

# 2.7 REDUNDANCY PAY

# 2.7.1 What entitlement will I receive if my position is made redundant?

If your position is made redundant, you will receive a severance payment of 2.5 weeks for each complete year of service, to a maximum of 30 weeks in both voluntary and involuntary redundancy situations. Payments are calculated on the

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **13** of **53**  TRP at the time of termination. If you have at least 1 year's continuous service, but less than 2 years, you will be entitled to a severance payment of 4 weeks in accordance with the provisions of the National Employment Standards.

You will also receive payment for the following:

- unused accrued annual leave
- pro-rata unused long service leave according to the provisions of this Agreement.

The above payments will not be offset against accumulated superannuation benefits, which will be available in accordance with the trust deeds of the relevant Superannuation Fund.

# 2.7.2 What other entitlements apply to me in a redundancy situation?

ARTC will provide to you the following services in a redundancy situation upon approval of your manager:

- Outplacement services, counselling and assistance including advice on all entitlements, independent financial planning guidance, assistance to plan lifestyle and career strategies and assistance with job search techniques and interview skills;
- Reasonable paid leave will be provided to a redundant employee to attend job interviews. You will at the request of ARTC, produce proof of attendance at an interview.
- If you were to pass away within the notice period of termination for reasons of redundancy and you have a period of eligible service which would have entitled you to redundancy pay, such redundancy pay entitlement will be paid to your estate.

# **2.8 APPRENTICES**

### 2.8.1 What provisions apply to me as an Apprentice?

The terms of this Agreement will apply to apprentices, including adult apprentices, except where it is otherwise stated or where special provisions are stated to apply. Apprentices may be engaged in trades or occupations provided for in this clause where declared or recognised by an Apprenticeship Authority.

# 2.8.2 What apprenticeship trades are applicable under this Agreement?

Subject to the provisions of clause 2.8.1 an apprenticeship may be undertaken in any of the following trades:

(1) Engineering Tradesperson (Mechanical).

(2) Engineering Tradesperson (Fabrication).

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **14** of **53**  When at any time ARTC has a need to enter into a contract of training with a person, the apprenticeship and training (will insofar as the *Fair Work Act 2009*, as amended allows), be in accordance with the requirements of the Apprenticeship Authority or State legislation.

# 2.9 HOURS OF WORK

# 2.9.1 What are my ordinary hours of work over a 28 day period?

If you are a full-time employee, you will be rostered to work an average of 152 hours over 28 days. The span of ordinary hours is from 0600 to 1800 Monday to Friday (inclusive), except where rostered according to clause 2.10. Where, by mutual agreement, you commence your shift between 0430 and 0600, penalties will not apply.

Your hours of work will be continuous on any 1 day, excluding your meal breaks.

# 2.9.2 Am I entitled to rostered days off?

Yes, you may work your ordinary hours so that you get 1 rostered day off (RDO) each 4 weeks on a day subject to agreement with your manager.

Your RDO will be rostered adjacent to other non-working days, subject to ARTC's business requirements.

# 2.9.3 What if I am required to work on my RDO?

If ARTC requires you to work on a day that has been agreed to be your RDO, you will get an alternative day off as soon as is practical and by agreement with your manager.

# 2.9.4 Can I accumulate my RDOs and add them to my annual leave?

Yes. You may accumulate a maximum of 3 RDO's in any 1 year if your manager agrees and it does not affect the operation of the business. The accumulated RDO's can be taken together as a block, or added to your annual leave.

### 2.10 ROSTERING

# 2.10.1 Can ARTC roster me to work afternoons, nights, weekends and public holidays?

Yes. You may be required to work to a roster that is outside the hours of 0600 to 1800, Monday to Friday.

# 2.10.2 Can I be required to work at places other than my designated work location?

Yes. ARTC's business requirements may mean that you will be rostered to work away from your designated work location.

# 2.10.3 Will I be consulted about my roster?

Yes. ARTC will draft, discuss and implement your roster through consultation at a local level prior to the posting of your roster. Consultation will take account of personal circumstances, balanced with equitable sharing and business needs.

Rosters will be posted 21 days in advance and will only change in an emergency or in unforeseen circumstances. You will be consulted about any changes to your roster. The maximum number of times your shift can be changed is 3 shifts in a 28 day period.

Your rosters will take account of:

- Ordinary shifts to be worked by an employee will be of a maximum length of 12 hours;
- The maximum number of consecutive 12 hour shifts that may be rostered is 4;
- Ordinary shifts to be worked by an employee will be of a minimum length of 6 hours;
- Except by mutual agreement, you will not be rostered to work more than 2 weekends in any 28 day period (for the purposes of this clause only, 'weekend' means either a Saturday, a Sunday or both Saturday and Sunday);
- The maximum number of consecutive shifts you may be required to work is 14. In order to meet operational requirements, you may be rostered to work in locations that require performance of a 14/7 roster, which is comprised of 14 ordinary shifts. In this instance, your hours of work will be averaged over a three (3) month cycle, equivalent to 456 hours over 84 days.
- Weekend rosters will take consideration of available competencies and volunteers;
- You will not be rostered to work within 10 hours of finishing your last shift;

ARTC's rostering will be based on fatigue management principles which:

- Address the opportunity for quantity and quality of sleep, particularly the 'time of day' effect;
- Ensure the number of consecutive shifts (in particular night shifts), shift lengths and roster periods between shifts are considered in roster compilation; and,
- Understand that you have a need to balance the competing requirements of your job with your social and home life.

# 2.11 MEAL BREAKS

# 2.11.1 What meal breaks am I entitled to?

You will be rostered for an unpaid meal break of not less than 30 minutes during the course of an 8 hour shift, provided that it does not interfere with operational requirements. Where an unpaid meal break is provided, you should not be required, where practical, to work more than 5 hours without a break.

# 2.11.2 What other rest breaks am I entitled to?

If you work greater than 10 hours you will be allowed a rest break of 20 minutes without deduction of pay. A further 20 minute paid rest break will be provided after each subsequent 4 hour period. The rest break will be organised to ensure continuity of operations.

# 2.12 NOTICE

# 2.12.1 What period of notice applies where I am required to work at a location other than my designated work location?

In the situation where you are required to commence work at a location other than your designated work location, in the case of programmed work, you will be provided with a minimum of one week's notice of such requirement. For urgent work, a minimum of 48 hours' notice and in the event of derailment or emergency attendance to restore service, every endeavour will be made to provide at least 48 hours' notice except where it is not possible to do so.

With respect to clause 2.12.1 the parties agree to regularly monitor the application of these provisions through the consultative committee to ensure the intentions of the parties are observed.

# 2.13 ALLOWANCES

### 2.13.1 On-Call / Call Out

# 2.13.1.1 Will I be required to participate in on-call rosters and attend call outs?

Yes. You could be rostered to be on-call and to attend call outs as required, to meet ARTC's business requirements. Call-out rosters will take consideration of available competencies, volunteers and fatigue management principles.

# 2.13.1.2 What will I be paid if I am on-call?

If you are required by ARTC management to be available outside normal working hours for recall to work, you will be paid an allowance of \$25.50 per rostered day or shift, or \$51.00 when on call for a non rostered day or shift. If you are rostered to be on call, you must be contactable and available for duty when required.

The on-call allowance will be increased in line with the salary increases specified in 3.1.1.1 as detailed below:

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017

Page **17** of **53** 

Allowance type	On commencement of new agreement	12 months after commencement	24 months after commencement
On-call day	\$25.50	\$26.01	\$26.54
On-call day and night	\$51.00	\$52.02	\$53.08

# 2.13.1.3 What will I be paid for a call-out?

If you are recalled to work overtime after leaving the work location (whether notified before or after leaving the location), you will be paid for a minimum of four hours' work at the appropriate rate for each time so recalled; provided that, except in the case of unforeseen circumstances arising, you will not be required to work the full four hours if the job you were recalled to perform is completed within a shorter period. An employee who is paid an on-call allowance will be paid for a minimum of three hours' work for each time so recalled. This clause will not apply in cases where it is customary for an employee to return to ARTC's place of business to perform a specific job outside ordinary working hours, or where the overtime is continuous (subject to a reasonable meal break), with the completion or commencement of ordinary time.

Overtime worked in the circumstances specific in this clause will not be regarded as overtime for the purpose of clause 3.2.3 when the actual time worked is less than four hours on such recall or each of such recalls.

# 2.13.2 What applies to me if I am required to work on a Saturday?

An employee required to work overtime on a Saturday will be afforded at least four hours' work or paid for four hours at the appointed rate, except where such overtime is continuous with overtime commenced on the day previous.

# 2.13.3 What applies to me if I am required to work on a Sunday?

An employee required to work overtime on a Sunday will be afforded at least four hours' work or paid for three hours at the appointed rate, except where such overtime is continuous with overtime commenced on the day previous.

# 2.13.4 What about a rest break whilst working overtime?

An employee working overtime will be allowed a rest break of 20 minutes without deduction of pay after each four hours of overtime worked if the employee continues work after such rest time.

Where a day worker is required to work on a Saturday or on a rostered day off, the first prescribed rest time will, if occurring between 10:00am and 1.00pm, be paid at ordinary rates.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **18** of **53**  Unless the period of overtime is less than one and a half hours, an employee before starting overtime after working ordinary hours, will be allowed a rest break of 20 minutes which will be paid for at ordinary rates. ARTC and the employee may agree to any variation of this provision to meet the circumstances of the work in hand provided that ARTC will not be required to make any payments in respect of any time allowed in excess of 20 minutes.

# 2.14 SHIFT WORK

# 2.14.1 What is shift work?

For the purposes of this agreement, "afternoon shift" means any shift finishing between 1800 hours and midnight (inclusive); and "night shift" means any shift finishing after midnight or a shift that commences between midnight and 0559 (inclusive).

# 2.14.2 What will I be paid for working an "afternoon shift"?

If you are rostered to work an "afternoon shift," you will be paid an additional 20% of your hourly rate based on your nominal salary rate for all ordinary hours worked where the shift is not subject to an equal or higher payment.

# 2.14.3 What will I be paid for working a "night shift"?

If you are rostered to work a "night shift," you will be paid an additional 25% of your hourly rate based on your nominal salary rate for all ordinary hours worked where the shift is not subject to an equal or higher payment.

# 2.14.4 What will I be paid if I am rostered to work on a Saturday or Sunday or Public Holiday?

If you are rostered to work on a Saturday, you will be paid at 150% of your nominal salary rate, if you are rostered to work on a Sunday, you will be paid at 200% of your nominal salary rate, and, if you are rostered to work on a public holiday, you will be paid at 250% of your nominal salary rate.

# 2.14.5 What if I work at least five successive shifts?

If you are a shift worker and you work on any afternoon or night shift which does not continue for at least five successive afternoons or nights, you will be paid at the rate of time and a half for all ordinary time occurring during such shifts.

On completion of the fifth consecutive afternoon or night's work, you will be deemed to have been employed on afternoon or night shift as the case may be, during the preceding four afternoons or nights, and thereafter during any subsequent consecutive afternoon or nights you are so employed. The sequence of shift work will not be deemed to be broken under this paragraph by reason of the fact that the works are erased on a Saturday, Sunday or any public holiday.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017

# 2.14.6 What if I work an extended night shift?

If you:

- during a period of engagement on shift, work night shift only; and
- remain on a night shift for a longer period than four successive weeks;

You will be paid 30% more than the ordinary rate for all time worked during ordinary working hours on such night shifts in excess of four weeks.

# 3. REMUNERATION AND RELATED

# 3.1 CLASSIFICATION AND SALARY STRUCTURE

# 3.1.1 What is the salary structure at ARTC?

You will be appointed to one of the classifications below, based on your achievement of the appropriate competencies for the level according to the competency profiles at Appendix 1. For the purposes of this Agreement, your salary level is based on the Total Remuneration Package (TRP) shown in clause 3.1.2.1 for Non-Supervisory Infrastructure Maintainers. Work Group Leaders and Team Leaders will be appointed at the relevant TRP amount as specified in clause 3.1.2.2. Apprentices will be appointed at the relevant TRP amount as specified in clause 3.1.2.3. These rates are inclusive of the 2% salary increase specified in clause 3.1.1.1

# 3.1.1.1 When are salary increases applicable?

During the nominal term of this Agreement you will receive a 2% salary increase on the first full pay period on or after the following dates:

- Upon commencement of the new Agreement
- 12 months after the commencement of the new Agreement
- 24 months after the commencement of the new Agreement

Increases to the TRP amounts are inclusive of applicable Superannuation Guarantee Contribution increases, to the extent the increases are legislated during the term of this Agreement.

# 3.1.2 What salary increases will I receive over the term of this Agreement?

Infrastructure Maintainer Level	TRP 2017	TRP 2018	TRP 2019
Level 1	\$54,830	\$55,927	\$57,046
Level 2	\$60,777	\$61,993	\$63,233
Level 3	\$67,251	\$68,597	\$69,969
Level 4	\$70,516	\$71,927	\$73,366
Level 5 (by appointment)	\$77,430	\$78,979	\$80,559

### 3.1.2.1 Non-Supervisory Staff:

# 3.1.2.2 Supervisory Staff:

Work Group	TRP	TRP	TRP
Leader	2017	2018	2019
	\$90,770	\$92,586	\$94,438

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **21** of **53** 

Team Leader	TRP	TRP	TRP
	2017	2018	2019
	\$93,394	\$95,262	\$97,168

# 3.1.2.3 I'm an Apprentice, what is my salary?

Apprentices will be paid a yearly salary as per the table below:

Apprentice (Year)	TRP 2017	TRP 2018	TRP 2019
Year 1	\$35,182	\$35,886	\$36,604
Year 2	\$41,645	\$42,478	\$43,328
Year 3	\$52,455	\$53,505	\$54,576
Year 4	\$60,374	\$61,582	\$62,814

# 3.1.3 What is included in my Total Remuneration Package?

Your Total Remuneration Package is an annualised salary that includes:

- Annual leave loading;
- Superannuation Guarantee Contributions;
- All allowances; and
- A requirement to work at locations other than your designated work location.

Your Total Remuneration Package *does not* include:

- Overtime;
- Shift, weekend and public holiday rates;
- On-call and call-out payments; or
- Living away from home expenses.

# 3.1.4 What is the Competency Structure for Non-Supervisory Staff at ARTC?

You will be appointed to one of the levels in the salary structure at clause 3.1.2.1 based on your achievement of the appropriate competencies as listed in Appendix 1.

If you are a Work Group Leader or Team Leader, you will be appointed in accordance with clause 3.1.2.2.

# 3.1.4.1 How do I apply for competency assessment?

Once you have completed all necessary log books, where required, and/or training for the competencies you want to be assessed against, you need to provide them to your Work Group Leader and Area Manager. They are to sign the log books to indicate you are ready for assessment. The Area Manager will then request from the relevant People representative for an assessment to take place.

If you are not assessed within 3 months of the request, once assessed competent, your competency progression will be effective from the date 3 months post the request for that assessment.

All other competency progression outcomes as a result of an assessment conducted within 3 months of the request, will be effective from the date of the successful assessment.

# 3.1.4.2 What if I don't have my log book (where required) and/or training records when I am due to be assessed?

You will not be assessed and you will not be entitled to back pay to this date. You must follow the correct procedure above to be entitled to any back pay for competency progression outcomes.

# 3.2 OVERTIME

# 3.2.1 Will I be required to work overtime?

Yes. ARTC may require you to work reasonable overtime at overtime rates and you will work overtime in accordance with such requirements.

### 3.2.2 What compensation will I receive for working overtime?

For all work done outside the nominated ordinary hours or spread of hours on Monday to Saturday, the rate of pay will be 1.7 times the ordinary time rate of pay. All overtime performed between midnight Saturday and midnight Sunday will be paid at double time.

Except as provided in this clause or clause 3.2.3 hereof in computing overtime, each day's work will stand alone. Day will mean all the time between the commencing time of one day and the commencing time of the next succeeding day.

For the purposes of this clause, ordinary hours will mean the hours worked in accordance with clause 2.9 of this Agreement.

All overtime is calculated on your nominal salary rate. The hourly rate, when computing overtime, will be determined by dividing the appropriate weekly rate by 38, even including in cases when an employee works more than or less than 38 ordinary hours in a week.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page 23 of 53

# 3.2.3 What is the rest period after working overtime?

When overtime work is necessary it will, whenever reasonably practicable, be so arranged that you have at least 10 consecutive hours off duty between the work of successive days.

If you (excluding casual employees) work so much overtime between the termination of ordinary work on one day and the commencement of your ordinary work on the next day and you have not had at least 10 consecutive hours off duty between those times you will, subject to this clause, be released after completion of such overtime until you have had ten consecutive hours off duty without loss of pay for ordinary working time occurring during such absence.

# 3.2.4 What if I'm required to work without having had 10 consecutive hours off?

If on the instructions of ARTC you resume or continue work without having had such 10 consecutive hours off duty, you will be paid at double rates until released from duty for such period and will then be entitled to be absent until you have had 10 consecutive hours off duty without loss of pay for ordinary working time occurring during such absence.

The provisions of this clause will also apply in the case of shift workers.

It is agreed that the ten hour period referred to herein can be reduced to eight hours in the event that the reason arises consequent upon a derailment (which is expected to be repaired for a turnaround of less than 3 days), and then only in circumstances where the health and safety of the employee is not compromised as a result of the reduction.

# 3.2.5 What transport arrangements apply after working overtime?

When an employee, after having worked overtime or a shift for which they have not been regularly rostered, finishes work at a time when reasonable means of transport are not available, ARTC will re-imburse the employee for the cost of conveying the employee to their usual place of residence or to the nearest appropriate public transport. In the event that ARTC provides this conveyance, re-imbursement will not apply.

# 3.3 TRAVELLING AWAY FROM HOME

# 3.3.1 Can I be required to work at locations which will temporarily require me to travel away from home and stay overnight?

Yes.

# 3.3.1.1 What happens if I am required to travel away from home and stay overnight?

If you are regularly rostered (e.g. migratory and regional teams), or if you are otherwise required to work at a location that does not permit you to return to your declared place of residence (home) each night, you will be compensated by ARTC.

# 3.3.1.2 What will I be paid when I am required to travel away from home and stay overnight?

If you are required to travel away from home and stay overnight, you will be paid a daily expense payment with your fortnightly salary payment on the following basis:

- Breakfast \$20.40
- Lunch \$23.46
- Dinner \$40.80
- Accommodation \$83.64

as a total payment of \$168.30.

OR

At your request, ARTC will pay your meals and accommodation directly to the supplier of the services.

The daily expense payment will be increased in line with the salary increases specified in 3.1.1.1 as detailed below:

Expense type	On commencement of new agreement	12 months after commencement	24 months after commencement
Breakfast	\$20.40	\$20.81	\$21.23
Lunch	\$23.46	\$23.93	\$24.41
Dinner	\$40.80	\$41.62	\$42.45
Accommodation	\$83.64	\$85.32	\$87.02
Total	\$168.30	\$171.68	\$175.11

# 3.4 CARRYING OUT HIGHER DUTIES

# 3.4.1 How will I be paid if I am carrying out higher duties?

If you are required to carry out higher duties at *Level 5* or as a *Work Group Leader or Team Leader*, you will be paid the salary applicable to the higher position for each full shift of acting in that position.

# 4. BALANCING WORK AND LIFE

### **4.1 ANNUAL LEAVE**

# 4.1.1 What is my entitlement to Annual Leave?

A period of 28 consecutive days, exclusive of any public holidays prescribed under clause 4.6 occurring during the period will be given and taken as leave annually to all employees, other than casual employees, which accrues progressively during the year.

# 4.1.1.1 I work shift work and weekends, do I get additional annual leave?

If you are rostered to work an average of 2 weekends (Saturday *and* Sunday) out of 4 over a 3 month period, you will receive an additional 1.25 days paid annual leave to a maximum of 1 additional week per 12 month period.

# 4.1.2 Under what methods can I take Annual Leave?

You may take either 28 consecutive days, or two separate periods of not less than seven consecutive days in all cases exclusive of any public holidays occurring therein, will be given and taken within twelve months from the date when the right to annual leave accrued. By agreement between ARTC and the employee concerned and the employee providing a minimum of two weeks' notice to ARTC of their request for leave, annual leave may be taken in single day absences of up to a maximum of five days in any one year.

Where an employee requests that annual leave be allowed in one continuous period, such request will not be unreasonably refused. In the event of lack of agreement between the parties, the matter will be referred to the Dispute Settlement Procedure for determination.

In the circumstances where a public holiday falls within one day of a weekend or another public holiday, the provision of clause 4.1.9 may be altered by agreement between ARTC and a majority of employees affected under this Agreement, to provide that a day of annual leave entitlement may be granted on the day between the said public holidays and/or weekend if an employee, or ARTC, requests it.

Where annual leave is proposed to be given and taken in two periods, one of which is to be given in conjunction with the Christmas and New Year holidays,

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **26** of **53**  representatives of ARTC and employees will meet and agree the commencing and finishing dates for the following Christmas/New Year period of leave. Where no agreement can be reached between the representatives, the matter will be referred to the Dispute Settlement Procedure for determination.

# 4.1.3 How much leave can I accrue?

You can save (accrue) up to 4 weeks annual leave. If you wish to accrue more than 4 weeks annual leave, you must obtain agreement with your manager.

# 4.1.4 Can I cash out my annual leave without taking time off?

Yes. Your request will need to be in writing.

Your request for paid annual leave to be cashed out must not result in your remaining accrued annual leave entitlement being less than 4 weeks.

Each request to cash out annual leave must be by a separate written agreement between you and ARTC.

You will be paid the full amount that would have been paid had you taken the annual leave.

# 4.1.5 What proportionate leave applies on termination?

If your employment ends, any untaken annual leave owing to you will be paid to you in your final pay.

# 4.1.6 What happens if there is a break in my service?

Where you break the continuity of service by absence from work for any reason other than a reason set out in clause 4.1.7, the amount of leave which you would have been entitled under clause 4.1.1 will be reduced by 1/48 for each week or part thereof during which any such time absence occurs and the amount of payment in lieu of leave to which any such absence occurs and the amount of payment in lieu of leave to which you would have been entitled under clause 4.1.8 will be reduced by 1/12 of a week's pay for each week or part thereof during which any such absence occurs.

No reduction will be made in respect of any absence unless ARTC informs you in writing of ARTC's intention to do so within fourteen days of the termination of absence.

# 4.1.7 How is continuous service calculated?

For the purposes of this clause, service will be deemed to be continuous notwithstanding:

- (a) Any interruption or determination of the employment by ARTC if such interruption or determination has been made merely with the intention of avoiding obligations hereunder in respect of leave of absence;
- (b) Any absence from work on account of personal sickness or accident or on account of leave lawfully granted by ARTC; or
- (c) Any absence with reasonable cause, proof whereof will be upon the employee.

In cases of personal sickness or accident, or absence with reasonable cause, you will become entitled to the benefit of this clause. You will inform the company, in writing if practicable, within 24 hours of the commencement of such absence of your inability to attend for duty and as far as practicable the nature of the illness, injury or cause and the estimated duration of your absence.

# 4.1.8 What will I be paid whilst on leave?

You will be paid at the rate attributable to your total annual remuneration package while on annual leave.

# 4.1.9 Is there an Annual Close Down?

Notwithstanding anything contained in this Agreement if leave is granted in conjunction with the Christmas/New Year holidays ARTC may either:

- Stand off without pay during the period of leave, any employee who has not yet qualified under clause 4.1.1;
- Stand off for the period of leave any employee who has not qualified under clause 4.1.1 hereof and pay them (up to the period of the leave then given) at a rate of 1/12 of an ordinary week's wages in respect of each five days continuous service (excluding overtime);
- Provided that where ARTC decides to close down the establishment at the Christmas/New Year period for the purpose of giving the whole annual leave due to all, or the majority of employees qualified for such leave, at least two months' notice of the intention to do so will be given to the employees.

# 4.1.10 When does leave commence when I'm on a distant job?

If you are still engaged on a distant job when annual leave is granted and you return by the first reasonable means of transport to your designated work location, annual leave will commence on the first full working day following your return to your designated work location.

# 4.1.11 Do I need to take leave?

The annual leave provided by this clause must be taken as leave and except as provided by clause 4.1.4, payment will not be made or accepted in lieu of annual leave.

Annual leave will be given at a time fixed by ARTC within a period not exceeding six months from the employee anniversary date.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **28** of **53** 

# 4.1.12 What if I am sick whilst on annual leave?

If you are sick when you are on annual leave, you may apply to substitute sick leave for annual leave if you provide your manager with a valid medical certificate.

# 4.1.13 How do I notify of my annual leave preferences in advance?

You will need to post your leave preferences at least 12 months in advance, between 1 November and 31 December each year. Your manager will post approvals within a month to inform you if your annual leave preference is available.

# 4.1.14 What if I don't post my annual leave preferences in advance?

If you do not indicate a preference for annual leave, your manager will assign the dates of your annual leave.

The objective of the process is to help ensure that a minimum of two weeks annual leave can be taken by employees with school age children within a recognised school holiday period and to ensure the equity in the allocation of annual leave rostering.

# 4.2 PERSONAL LEAVE INCORPORATING SICK LEAVE AND CARER LEAVE

# 4.2.1 What is my personal/carer leave entitlement?

You are entitled to 10 days personal/carer leave per year in accordance with the *National Employment Standards.* Any unused personal/carer leave accrues.

If an employee has used their entitlement to paid carer's leave under the *Act*, then they may take up to two days unpaid carer's leave in accordance with the *Act*.

# 4.2.2 What is carer leave?

Carer leave is paid leave provided to you for the purposes of caring for an immediate family or household member who requires you to provide them care and support due to an illness, injury or an unexpected emergency affecting them.

# 4.2.2.1 What does "immediate family or household member" mean?

Members of your immediate family or household include:

- Your spouse (including former spouse, de-facto spouse and former de-facto spouse). A de-facto spouse means a person who lives with you on a bona fide domestic basis; and
- A child or an adult child (including an adopted child, stepchild or ex-nuptial child), parent, grandparent, grandchild or sibling, of you or your spouse.

# 4.2.3 What is personal leave?

Personal leave is paid leave taken when you are not fit for work because of a personal illness or personal injury.

# 4.2.4 How is personal/carer leave used?

Any personal/carer leave taken during a year will be debited against your accrued balance of personal/carer leave up to the maximum of your accrued balance.

You should advise ARTC in your leave application whether you are taking personal or carer leave.

# 4.2.5 What about sick leave?

In addition to your personal/carer leave entitlement, you are also entitled to unlimited sick leave. Your unlimited sick leave entitlement will become available when all of your personal/carer leave entitlement has been used.

You must supply suitable evidence of an illness or injury in the following occasions: after two consecutive day's absence, after three uncertified days of sick leave within the service year, on a day before or after a rostered day off, weekend or public holiday, or on the first or last day of a rostered shift.

# 4.2.5.1 What do I do if I am sick or need to claim carer leave?

If you are unable to come to work due to personal illness or injury, or you are claiming carer leave, you should notify your manager within at least 1 hour of your rostered starting time and tell your manager when you expect to return to work. You must also provide evidence of your illness or injury, or the illness or injury of your immediate family or household member.

# 4.2.5.2 What happens if I am sick for a long period of time?

If you have a long-term illness that causes you to be absent for more than 6 months then, if directed by ARTC, you will attend an appointment with a medical practitioner nominated by ARTC to determine the likelihood of you returning to work. If the medical advice confirms that you will be unable to return to work, your employment may terminate when a copy of that medical advice is given to you.

# 4.3 COMPASSIONATE AND BEREAVEMENT LEAVE

# 4.3.1 What are compassionate and bereavement leave and to what extent am I covered?

On each occasion on which a member of your "immediate family or household":

• Contracts or develops a personal illness, or sustains a personal injury, that poses a serious threat to his or her life, you may take paid leave of up to 2 days; or

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **30** of **53**  • Dies, you may take paid leave of up to 3 days paid bereavement/compassionate leave.

You must provide ARTC with satisfactory evidence of the illness, injury or death of the member of your family or household in order to receive payment for the leave taken.

# 4.3.2 What if I need leave to cover bereavement or compassionate circumstances for someone close to me but they are not an immediate family or household member?

If a person close to you dies, but is not part of your "immediate family or household", ARTC may allow you time off to attend the funeral by agreement with your manager.

# **4.4 PARENTAL LEAVE**

Subject to the terms of this clause, employees are entitled to maternity, paternity and adoption leave in connection with the birth or adoption of a child in accordance with section 70 of the *Fair Work Act 2009*.

# 4.5 OTHER LEAVE

# 4.5.1 What leave am I entitled to if I am required to serve on a jury (jury duty)?

If you are required to attend for jury service you will be reimbursed by ARTC an amount equal to the difference between the amount paid in respect of such attendance for such jury service and the amount of wage that would have been paid in respect of the ordinary time worked had you not been on jury service.

You will notify ARTC as soon as practicable of the date upon which you are required to attend for jury service and will provide ARTC with proof of attendance, the duration of such attendance and the amount received in respect thereof.

# 4.5.2 Am I entitled to emergency services leave?

Yes. If you are a member of an emergency service (for example, Country Fire Service or State Emergency Services), you must advise your manager of the relevant details of your membership responsibilities. You may be released from duty, if called upon at times of declared emergencies, provided that you are not required to attend to the ARTC's assets. You should advise your manager immediately when notified that you are required for duty in an emergency.

When you return to work, you should provide proof of your attendance certified by an authorised representative of the emergency service to which you were attached. Wherever possible, times of attendance should be shown. Paid leave will be provided for such authorised absences.

# 4.5.3 Can I take military leave?

Yes. Subject to legislative requirements, military leave may be granted to you if you are a volunteer part-time member of the Australian Defence Forces. ARTC will provide you with up to 10 days paid leave per year, based on the difference between your ordinary pay and the payment you receive as reserve pay.

# 4.6 PUBLIC HOLIDAYS

# 4.6.1 What is my entitlement to public holidays?

You, (excluding a casual employee as defined), will be entitled to the following holidays without deduction of pay:

- New Year's Day, Good Friday, Easter Saturday, Easter Monday, Christmas Day and Boxing Day; and
- The following days as prescribed in the relevant states, territories and localities: Australia Day, Anzac Day, Queen's Birthday and Labour Day; and
- One other day to be specified according to state or locality.

# 4.6.2 What if the public holidays fall on a weekend?

When the following public holidays fall on a Saturday or Sunday, a holiday in lieu will be observed:

- Christmas Day: a holiday in lieu will be observed on the 27<sup>th</sup> or 28<sup>th</sup> of December;
- Boxing Day: a holiday in lieu will be observed on the 28<sup>th</sup> of December;
- New Year's Day: a holiday in lieu will be observed on the next Monday.

Where in a state, or locality, public holidays are declared or prescribed on days other than those set out in clause 4.6.1, those days will constitute additional holidays for the purpose of the Agreement.

# 4.6.3 Can public holidays be substituted?

Yes. ARTC and the employees may agree to substitute another day for any prescribed day in this clause. For this purpose, the consent of the majority of employees will constitute agreement.

An agreement pursuant to the above provisions of clause 4.6.2 will be recorded in writing and be available to every affected employee.

# 4.6.4 What will I be paid if I am required to work on a public holiday?

If you are required to work on a public holiday, you will be paid at the appropriate rate for all hours worked with a minimum payment of four hours at the appropriate rate.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **32** of **53**  Whenever any holiday falls on an employee's ordinary working day and you are not required to work on such day, you will be paid for the ordinary hours that you would have worked on such day had it not been a holiday.

If you are rostered to work on a public holiday, you will be paid at 250% of your nominal salary rate in accordance with clause 2.14.4.

If a public holiday, as defined in clause 4.6.1 falls on a week day within an employee's period of annual leave and is observed on a day which in the case of that employee would have been an ordinary working day, there will be added to that period of leave one day being an ordinary working day for each such holiday observed aforesaid.

All holidays are to be computed in a manner consistent with the adopted method of implementation of rostering in accordance with clause 2.10.

If you return to your designated work location or finish a shift at the designated work location, not later than 0400 hours on any public holiday and you are not again booked for that day, you will be treated as having had a paid holiday.

If you are off duty owing to leave without pay or sickness, including accidents on or off duty, except time for which you are entitled to claim sick pay, any holiday falling during such absence will not be treated as a paid holiday. In the instance, however, you are on or are available for duty on the working day immediately preceding a paid holiday or resume or are available for duty on the working day immediately following a holiday, you will be entitled to a paid holiday on such holiday.

Unless at your own request, you will not be booked off for a holiday at a foreign or a temporary designated work location.

Any holiday occurring during the period in which you are on long service leave will be calculated as portion of the long service leave and extra days in lieu will not be granted.

# 4.7 ARE CASUAL EMPLOYEES ENTITLED TO ANY PAID HOLIDAYS UNDER THIS AGREEMENT?

No.

# 4.8 LONG SERVICE LEAVE

### 4.8.1 What is my entitlement to Long Service Leave?

You will accrue an entitlement to Long Service Leave in accordance with the *South Australian Long Service Leave Act 1987.* 

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **33** of **53**
# 4.8.2 Can I cash in Long Service Leave?

Yes. You can cash in a portion of your Long Service Leave entitlement, in line with the provisions of the *South Australian Long Service Leave Act 1987*.

# 5. HEALTH WELLBEING AND ASSESSMENT

# 5.1 HEALTH ASSESSMENT

#### 5.1.1 When do I have to have a medical?

As an employee of ARTC, you must comply with ARTC's requirements under the *National Standard for Health Assessment of Rail Safety Workers* including attending medical assessments.

#### 5.1.2 Do I have to pay for these Health Assessments?

No. ARTC will pay the cost of all medical tests, including specialist tests undertaken at ARTC's direction, up to the time you are determined fit or unfit for duty.

#### 5.1.3 When do I have to have my health assessment?

It is ARTC's intention that initial and follow up health assessments are conducted during normal working hours.

#### 5.1.4 What happens to me if I am deemed unfit for work?

If your health assessment determines that you are temporarily unfit for your normal duties, ARTC will endeavour to provide you with suitable alternate duties. Where alternate duties are available, you will continue to be paid your total remuneration package for a period of up to 3 months provided there is an expectation that you will return to your full duties.

#### 5.2 WORK CLOTHING AND PERSONAL PROTECTIVE EQUIPMENT (PPE)

#### 5.2.1Will I be given clothes and PPE for work?

Yes, ARTC will provide you with work clothes and safety gear as detailed in clause 5.2.4.

#### 5.2.2 Who is responsible for my issued work clothes and PPE?

You must correctly use and take reasonable care (including laundering) of all PPE and/or clothing provided to you by ARTC. Employees must at all times use the relevant PPE where a safety requirement exists.

# 5.2.3 What do I do if I need replacements for worn, defective or damaged work and safety gear?

If any of your issued work clothes or PPE are worn, defective or damaged, you should immediately report this to your manager or supervisor who will arrange replacements for you. Replacements will be on a fair wear and tear basis. Lost work clothing items and PPE will be replaced at ARTC's discretion.

#### 5.2.4 What clothes and safety gear will I be given?

You will receive allocated work clothing which incorporates the ARTC logo where appropriate. The allocated work clothing is the only acceptable form of clothing for infrastructure maintenance employees. The allocated work clothing will include:

- a) 2 pairs of safety footwear;
- b) 5 shirts & 5 pairs of pants or shorts, or action back overalls, or coveralls (ARTC may increase this allocation based on your work patterns. For example, if you regularly work a roster pattern of 8 days on, 6 days off, you may be allocated 8 shirts and 8 pairs of shorts or pants etc.);
- c) 1 jacket or 1 pullover;
- d) 1 hat; and
- e) Any other safety gear necessary to do your job.

#### 5.2.5 Who is responsible for washing my PPE and work clothes?

Laundering of work clothing is your responsibility unless ARTC policies provide otherwise.

# 5.2.6 Do I get to keep the work clothes and safety gear?

No. PPE and work clothing will remain the property of ARTC.

#### **5.3 INCLEMENT WEATHER**

# 5.3.1 What does "Inclement Weather" Mean?

"Inclement weather" will mean the existence of rain or abnormal climatic conditions (whether they be those of hail, cold, high wind, severe dust storm, extremes of high temperature or the like of any combination thereof), by virtue of which it is either not reasonable or safe for employees exposed, to continue working whilst such inclement weather prevails.

#### 5.3.2 How is an inclement weather situation managed?

If, owing to inclement weather conditions you are unable to work, you will be entitled to be paid for the time lost. It will be the sole responsibility of the employee in charge on the site of the work being carried out to determine whether or not and when work will cease owing to inclement weather conditions.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **35** of **53**  There will be no deduction of wages for time lost during inclement weather provided that any employee ceasing work of their own volition will not be entitled to payment for the time lost.

#### 6. RECOGNISING YOUR CONTRIBUTION

# 6.1 TRAINING AND DEVELOPMENT

#### 6.1.1 Will ARTC provide opportunities for learning and career development?

Yes. Recognising the performance focus of the business, opportunities for training will be provided by ARTC, subject to your manager's consideration of:

- your training needs and/or requirements;
- practical organisational impacts;
- any reasonable requests by you for development or change; and/or
- progression to the next level (up to level 4), will be by natural progression once you have achieved the competencies of that level in line with the competencies detailed in Appendix 1 which enables the attainment of nationally recognised qualifications.

The combination of competencies required in the different work locations will vary in accordance with the operational needs of that location.

ARTC is committed to ensuring employees are provided with genuine developmental opportunities for progression within their classification stream up to level 4. Typically an employee is reasonably expected to meet the requirements of level 4 within 4 years. ARTC and employees have a shared responsibility for competency achievement. Employees' advancement through the levels will be dependent upon the employee's willingness to actively participate in competency attainment, their capacity to competently perform the work and successfully meet all competency assessment requirements.

Advancement to higher levels (level 5) will result from the requirement / needs of the organisation for skills within the classification levels, the capacity of the employee to competently perform the work and the availability of positions. Progression to level 5 in each of the streams will be by appointment.

Individual Annual training plans will be developed through consultation with each employee to establish a plan for the year ahead. The training plans will indicate particular units of competence from the employee's respective classification level which are in line with ARTC's business needs, and allow for the opportunity for career progression. Employees will be provided with access to a copy of their individual training plan.

ARTC acknowledges that no employee will have their current level of pay downgraded as a result of the introduction of the new competency structure.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **36** of **53**  Upon the introduction of the new competency structure individual training plans will include all required competencies at or below the employee's current classification level, along with any competencies agreed between the employee and their manager. Employees will be required to complete the units of competence at or below their classification level prior to further progression occurring.

#### 6.1.2 How do I get access to training opportunities?

You are required to participate in the development of ARTC's Personal Development Plans through which you and your manager will plan your training.

## 6.2 BONUSES

# 6.2.1 Can I participate in ARTC's Bonus Plan?

Yes. If you are a permanent employee, you will be entitled to participate in the ARTC Bonus Plan, as amended from time to time.

# 7. CONSULTATION AND DISPUTE RESOLUTION

# 7.1 WILL I BE CONSULTED ON ANY MAJOR CHANGES HAPPENING AT ARTC?

Yes, where ARTC

(a) has made a definite decision to introduce a major change to production, program, organisation, structure or technology in relation to its enterprise that is likely to have a significant effect on the employees;

(b) or proposes to introduce a change to the regular roster or ordinary hours of work of employees, this clause will apply.

## **Major Change**

7.1.1 For a major change referred to in paragraph 7.1 (a):

(a) the employer must notify the relevant employees of the decision to introduce the major change and

(b) clauses 7.1.1.2 to 7.1.1.8 apply.

7.1.1.2 The relevant employees may appoint a representative for the purposes of the procedures in this term.

#### 7.1.1.3 lf

(a) a relevant employee appoints, or relevant employees appoint, a representative for the purposes of consultation; and

(b) the employee or employees advise the employer of the identity of the representative;

The employer must recognise the representative.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **37** of **53**  7.1.1.4 As soon as practicable after making its decision, the employer must:

7.1.1.4 (a) discuss with the relevant employees:

7.1.1.4 (a) (i) the introduction of the change; and

7.1.1.4 (a) (ii) the effect the change is likely to have on the employees: and

7.1.1.4 (a) (iii) measures the employer is taking to avert or mitigate the adverse effect of the change on the employees; and

7.1.1.4 (b) for the purposes of the discussion - provide, in writing, to the relevant employees:

7.1.1.4 (b) (i) all relevant information about the change including the nature of the change proposed; and

7.1.1.4 (b) (ii) information about the expected effects of the change of the employees; and

7.1.1.4 (b) (iii) any other matters likely to affect the employees.

7.1.1.5 However, the employer is not required to disclose confidential or commercially sensitive information to the relevant employees.

7.1.1.6 The employer must give prompt and genuine consideration to matters raised about the major change by the relevant employees.

7.1.1.7 If a term of this Agreement provides for a major change to production, program, organisation, structure or technology in relation to the enterprise of the employer, the requirements set out in paragraph 7.1.1(a) and clauses 7.1.1.2 and 7.1.1.4 are taken not to apply

7.1.1.8 In this term a major change is *likely to have a significant effect on employees* if it results in:

7.1.1.8(a) the termination of the employment of employees: or

7.1.1.8(b) major change to the composition, operation or size of the employer's workforce or to the skills required of employees; or

7.1.1.8(c) the elimination or dimunition of job opportunities (including opportunities for promotion or tenure); or

7.1.1.8(d) the alteration of hours of work; or

7.1.1.8(e) the need to retrain employees; or

7.1.1.8(f) the need to relocate employees to another workplace; or

7.1.1.8(g) the restructuring of jobs.

#### Change to regular roster or ordinary hours of work

7.1.2 For a change referred to in 7.1(b):

7.1.2(a) the employer must notify the relevant employees of the proposed change; and

7.1.2(b) clauses 7.1.2.1 to 7.1.2.5 apply.

7.1.2.1 The relevant employees may appoint a representative for the purposes of the procedures in this term.

7.1.2.2 lf:

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **38** of **53**  7.1.2.2(a) a relevant employee appoints, or relevant employees appoint a representative for the purposes of consultation; and

7.1.2.2(b) the employee or employees advise the employer of the identity of the representative

the employer must recognise the representative.

7.1.2.3 As soon as practicable after proposing to introduce the change, the employer must:

7.1.2.3(a) discuss with the relevant employees the introduction of the change; and

7.1.2.3(b) for the purposes of the discussion – provide to the relevant employees:

7.1.2.3(b) (i) all relevant information about the change, including the nature of the change; and

7.1.2.3(b) (ii) information about what the employer reasonably believes will be the effects of the change on the employees; and

7.1.2.3(b) (iii) information about any other matters that the employer reasonably believes are likely to affect the employees; and

7.1.2.3(c) invite the relevant employees to give their views about the impact of the change (including any impact in relation to their family or caring responsibilities).

7.1.2.4 However, the employer is not required to disclose confidential information or commercially sensitive information to the relevant employees.

7.1.2.5 The employer must give prompt and genuine consideration to matters raised about the change by the relevant employees.

7.1.2.6 in this term "relevant employees" means the employees who may be affected by a change referred to in clause 7.1.

# 7.2 CONSULTATION PROCESS

# 7.2.1 What is ARTC's Consultation Process

We recognise that communication and participation are essential elements for the effective operation of ARTC.

ARTC will consult directly and/or through established committee structures, with employees and any persons nominated by an employee or employees to represent their interests.

We will convene a Consultative Committee (CC) designed to provide for the exchange of information and consultation.

In addition to the CC structures, other communication and consultative mechanisms may be developed to encourage workforce participation in the decision making process.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **39** of **53** 

# 7.3 CONSULTATIVE COMMITTEE

## 7.3.1 What is the purpose of the Consultative Committee?

A Consultative Committee will be formed to assist management in the implementation of Best Practice initiatives consistent with the aims of this Agreement.

# 7.3.2 What is the composition of the Consultative Committee?

The Consultative Committee will consist of two (2) representatives of the workforce, elected by the workforce and two (2) representatives of management or such other equal numbers as agreed, who will meet as frequently as the Committee determines is necessary. It is intended that the members of the Committee would be from a broad cross-section of the workforce who are appropriately trained to undertake their responsibility.

The Consultative Committee will be required to keep minutes and to provide an agenda for each meeting.

In an effort to provide a new input to the Committee from time to time, the Committee may invite other representatives to attend the Consultative Committee meetings. Such representatives may be from subcontracting companies who participate in the maintenance operations, or a representative of ARTC's management team.

# 7.3.3 What are the scope of tasks undertaken by the Consultative Committee?

The tasks to be undertaken by the Consultative Committee will include, but will not be limited to the following:

- Monitor and review the implementation of this Agreement and other local issues, with the exception of Work Health and Safety (WHS) issues which will be referred to the relevant WHS committee;
- Review and discuss ARTC's business objectives and performance and future business plans. The committee will also identify and recommend proposals and initiatives to improve business performance.

# 7.4 TOOL BOX MEETINGS

#### 7.4.1 What are Tool Box Meetings and what is their purpose?

"Tool Box" meetings will be held monthly as a minimum for the purpose of discussing any matter concerning the general welfare of employees. Each meeting will be conducted by the individual supervisors and will involve each member of their immediate work team. Matters arising from the meetings, if not resolved at the time, will be processed through the relevant Work Group Leader.

# 7.5 DISPUTE SETTLEMENT PROCEDURE

7.5.1 What process do I follow to resolve a dispute in relation to my work with ARTC?

The following procedures will be used in order to prevent and settle disputes arising from this Agreement or in relation to the *National Employment Standards*.

The objectives of these procedures are to resolve disputes that arise under this Agreement or in relation to the *National Employment Standards* promptly, fairly and by direct consultation and/or negotiation at the workplace wherever possible.

Without prejudice to either party and except where a bona fide safety issue is involved, work will continue while matters in dispute are negotiated in good faith and in accordance with the procedures contained below.

ARTC and employees will attempt, wherever possible, to resolve the matter at the workplace level by taking the following steps as necessary:

- Step 1. The employee and a representative (at the employee's choice if requested) and their immediate manager will meet to try to resolve the matter. The relevant manager will endeavour to respond to the employee raising the matter within 48 hours, and advise of action being taken.
- Step 2. If the matter is not resolved, the parties will arrange further discussions involving more senior levels of management and a representative (at the employee's choice if requested). ARTC will endeavour to respond to the person raising the matter within 48 hours, and advise of action being taken.
- Step 3. If the matter is not resolved, further discussions will occur involving the employee and a representative (at the employee's choice if requested), and the employee's General Manager or nominee. ARTC will endeavour to respond to the person raising the matter within 48 hours, and advise of action being taken.
- Step 4. If the matter is not resolved, it may now be referred by either party to either:
  - (a) The Fair Work Commission; or

(b) An independent mediator where agreed and appointed by both parties.

Where the parties agree or where required by law, dispute resolution by the *Fair Work Commission*, or an independent mediator, will be binding.

It is understood that the parties to this procedure will not have recourse to the formal processes of the *Fair Work Commission* until they have endeavoured to resolve the issues between them in full accordance with the procedures contained in this Agreement.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **41** of **53** 

# 8. FLEXIBILITY TERM

# 8.1 INDIVIDUAL FLEXIBILITY ARRANGEMENT

### 8.1.1 Can the terms of this Agreement be varied?

Yes. ARTC and an employee covered by this Agreement may agree to make an Individual Flexibility Arrangement to vary the effect of terms stipulated below in 8.1.2 of this Agreement.

# 8.1.2 What is an Individual Flexibility Arrangement?

An Individual Flexibility Arrangement is a written Agreement genuinely agreed to between you and ARTC. An Individual Flexibility Arrangement can vary the clauses in this Agreement regarding:

- a) arrangements about when work is performed;
- b) overtime rates;
- c) penalty rates;
- d) allowances;
- e) remuneration; and
- f) leave and leave loading

ARTC will ensure that the terms of the Individual Flexibility Arrangement:

- (a) are about permitted matters under section 172 of the *Fair Work Act 2009*; and
- (b) are not unlawful terms under section 194 of the Fair Work Act 2009; and
- (c) result in you being better off overall than you would be if no arrangement was made.

#### 8.1.3 How do I make an Individual Flexibility Arrangement?

ARTC must ensure that the Individual Flexibility Arrangement:

- Is in writing
- Includes the name of ARTC and employee
- Is genuinely agreed
- Is signed by ARTC and employee and if the employee is under 18 years of age, signed by a parent or guardian of the employee;
- Includes details of:
  - The terms of the Agreement that will be varied by the arrangement;
  - How the arrangement will vary the effect of the terms:
  - How the employee will be better off overall in relation to the terms and conditions of his or her employment as a result of the arrangement; and
- States the day on which the arrangement commences.

ARTC will give you a copy of your Individual Flexibility Arrangement within 14 days of it being signed.

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017

# 8.1.4 How do I terminate (end) an Individual Flexibility Arrangement?

You may terminate your Individual Flexibility Arrangement by giving ARTC no more than 28 days written notice, or if ARTC and you agree in writing, at any time, after which time, your arrangements revert to the relevant provisions of this Agreement. If you wish to terminate your Individual Flexibility Arrangement with less than two weeks notice, you can do so by obtaining written agreement from ARTC.

# 8.1.5 Can ARTC terminate (end) my Individual Flexibility Arrangement?

Yes. ARTC may terminate your Individual Flexibility Arrangement by giving you no more than 28 days written notice, or if ARTC and you agree in writing, at any time, after which time, your arrangements will revert to the relevant provisions in this Agreement.

# Appendix 1 – ARTC South Australia Western Australia Infrastructure Maintenance Competency Structure

	Units	Unit Description
	TLIB1028	Maintain and Use Hand tools
	TLID1001	Shift materials using manual handling methods
	TLIE1003	Participate in basic workplace communication
Level 1	TLIF1001	Follow work health and safety procedures
All Streams	TLIF2010	Apply Fatigue Management Strategies
	TLIF2080	Safely Access the Rail Corridor
	TLIU2008	Apply Environmental procedures to rail infrastructure
	CPCCOHS1001A	Work Safely in the Construction Industry (White Card)
	TLIF1002	Conduct housekeeping activities
		Plus 3 electives from Electives level 1 and 2

	Units	Unit Description
	TLIB2085	Apply Track Fundamentals
	TLIB2092	Operate Minor Mechanical Equipment
	TLIS2030	Carry out track ballasting
	TLIS2031	Install Railway Sleepers
Level 2 Track	TLIS2034	Install and Repair Rail Fastening Systems
HACK	TLIS2044	Carry out rail installation
	TLIF2081	Perform Lookout Duties
	TLIF3083	Conduct track protection Assessment
		ARTCNTSA Part B - NCOP
	TLIB1093	Clean equipment and restore worksite
		Plus 3 electives from Electives level 1 and 2

	Units	Unit Description
	RIICM201D	Carry out measurements and calculations
	TLIE2007	Use Communication Systems
Level 3 Track	TLIF3003	Implement and monitor work health and safety procedures
TTACK	TLIJ3002	Apply Quality Systems
	RIICM203D	Read and Interpret plans and job specifications
	TLIL3065	Implement Track Occupancy Authority
	Plus	3 electives from Electives level 1 and 2 or 3 and 4

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **44** of **53** 

	Units	Unit Description
	Track Examination	
	TLIB3099	Examine Track Infrastructure
	TLIB3100	Visually Inspect Track Infrastructure
	TLIS2027	Install and maintain surface track drainage
	TLIE2008	Process workplace documentation
	TLIB3095	Check and Repair points and crossings
Level 4	TLIS3026	Implement track maintenance and construction
Track & Welding	TLIE2029	Conduct Workplace Information Briefings
5	TLIF2006	Apply accident-emergency procedures
	TLIB3094	Check and Repair Track Geometry
· .	Rail Testing	
	TLIS3010	Test rail using ultrasonic equipment
	TLIS3011	Test rail using non-destructive testing equipment
	TLIE2029	Conduct Workplace Information Briefings
	TLIF2006	Apply accident-emergency procedures
·····	TLIE2008	Process workplace documentation
	Plus 3 electives from	n Electives level 1 an 2 or 3 and 4 for Track Examination stream
	Plus 3 electives f	rom Electives level 1 and 2 or 3 and 4 for Rail Testing stream

	Units	Unit Description
	TL1B2085	Apply Track Fundamentals
	TLIB2092	Operate Minor Mechanical Equipment
	TLIS2030	Carry out track ballasting
	TLIS2031	Install Railway Sleepers
Level 2 Welding	TLIS2034	Install and Repair Rail Fastening Systems
weiumg	TLIS2044	Carry out rail installation
	TLIF2081	Perform Lookout Duties
	TLIF3083	Conduct track protection Assessment
		ARTC NTSA Part B - NCOP
	TLIB1093	Clean equipment and restore worksite
		Plus 3 electives from Electives level 1 and 2

	Units	Unit Description
	RIICM201D	Carry out measurements and calculations
	TLIE2007	Use Communication Systems
	TLIF3003	Implement and monitor work health and safety procedures
Level 3 Welding	TLIJ3002	Apply Quality Systems
	RIICM203D	Read and Interpret plans and job specifications
	TLIL3065	Implement Track Occupancy Authority
	TLIW2012	Grind Rails
	TLIW3015	Weld rail using aluminothermic welding process

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **45** of **53** 

TLIW3035	TLIW3035	Heat and cut materials using oxy-LPG equipment for the rail industry
	TLIF2006	Apply accident-emergency procedures
	Plus 3 electives from Electives level 1 and 2 or 3 and 4	

	Units	Unit Description
	TLIB2085	Apply Track Fundamentals
	TLIB1093	Clean equipment and restore worksite
	TLIS2030	Carry out track ballasting
Level 2	TLIB2001	Check and assess operational capabilities of equipment
Resurfacing	TLIC2059	Propel and operate light on-track equipment
	TLIB2091	Measure and record track geometry
	TLIF2081	Perform Lookout Duties
	TLIF3083	Conduct track protection Assessment
		ARTC NTSA Part B - NCOP
		Plus 3 electives from Electives level 1 and 2

	Units	Unit Description
	RIICM201D	Carry out measurements and calculations
	TLIE2007	Use Communication Systems
Level 3	TLIF3003	Implement and monitor work health and safety procedures
Resurfacing	TLIJ3002	Apply Quality Systems
	RIICM203D	Read and Interpret plans and job specifications
	TLIL3065	Implement Track Occupancy Authority
	TLIC2081	Pilot traffic within work on track authority limits
	Plus	3 electives from Electives level 1 and 2 or 3 and 4

	Units	Unit Description
	TLIS3039	Measure and mark track for resurfacing
	TLIE2008	Process workplace documentation
	TLIB3094	Check and repair track geometry
Level 4 Resurfacing	TLIE2029	Conduct Workplace Information Briefings
. Resurracing	TLIF2006	Apply accident-emergency procedures
	TLIC2058	Travel medium or heavy self-propelled on-track equipment
	TLIS3026	Implement track maintenance and construction
	Plus	3 electives from Electives level 1 and 2 or 3 and 4

	Units	Unit Description
	RIICM201D	Carry out measurements and calculations
Level 2 Structures	TLIB2084	Carry out routine maintenance of structures
511 40141 45	TLIB2086	Apply Awareness of structures fundamentals
	TLIB2092	Operate minor mechanical equipment
	TLIF2081	Perform Lookout Duties

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **46** of **53** 

T	LIF3083	Conduct track protection Assessment	
		ARTC NTSA Part B - NCOP	
R	RIIWHS202D	Enter and work in confined spaces	
F	RIIWHS204D	Work safely at heights	
Т	LIB1093	Clean equipment and restore worksite	
		Plus 3 electives from Electives level 1 and 2	

	Units	Unit Description	
	TLIE2007	Use communication systems	
	RIICCM203D	Read and interpret plans and specifications	
	TLIF3003	Implement and monitor work health and safety procedures	
	TLIJ3002	Apply Quality Systems	
	TLIS2013	Install Minor Structures	
	TLIS3039	Implement structures maintenance and installation of minor structures	
Level 3	TLIL3065	Implement Track Occupancy Authority	
Structures	TLIS2033	Install and repair temporary track supports	
	TLIW3035	Heat and cut materials using oxy-LPG equipment	
	TLIS2036	Use chemical repair products	
	AND Steel Structures Specialist Group		
	TLIB2082	Repair Steel Structures	
	TLIB2083	Maintain bridge bearings	
	TLIB2096	Repair concrete/masonry structures	
	OR Timber Structures Specialist Group		
	TLIB2081	Repair Timber Structures	
	TLIS2028	Install and replace transoms	
1	Plus	3 electives from Electives level 1 and 2 or 3 and 4	

	Units	Unit Description
	Steel Structures	
	TLIB3098	Examine concrete/masonry structures
	MEM18001C	Use hand tools
Level 4	MEM24001B	Perform basic penetrant testing
Structures	MEM24003B	Perform basic magnetic particle testing
	TLIE2008	Process workplace documentation
	TLIF2006	Apply accident-emergency procedures
	TLIE2029	Conduct Workplace Information Briefings
	TLIB3088	Examine steel structures
	TLIB3094	Check and repair track geometry

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **47** of **53** 

Timber Structures	
TLIB3098	Examine concrete/masonry structures
TLIB3087	Examine timber structures
TLIE2008	Process workplace documentation
TLIF2006	Apply accident-emergency procedures
TLIE2029	Conduct Workplace Information Briefings
TLIB3094	Check and repair track geometry
Plus	3 electives from Electives level 1 and 2 or 3 and 4

	Units Unit Description	
	Skills include:	
	These positions will be by merit selection as required by each of the corridors. Each position will have a position description developed. New Appointees to this level will be required to be competent up to and including Level 4 Installer/Certifier. Examples of areas of specification that fit into this level are -	
Level 5	<ul> <li>Coordinate ultrasonic testing by contractors;</li> </ul>	
Experienced Tradesperson /	- Computer based rail measurement and rail profiling;	
Specialist Track,	<ul> <li>Structures defect prevention and management;</li> </ul>	
Signals, Structures, Earthworks	– Rail wheel interface skills;	
	<ul> <li>Rail defect prevention and management;</li> </ul>	
	– Coordinate rail grinding;	
	<ul> <li>– Signals defect prevention and management;</li> </ul>	
	<ul> <li>Ability to define scope of works.</li> </ul>	
	Positions may also include plant fitters/mechanics	

	Units	Unit Description
	TLIB1028	Maintain and Use Hand tools
	TLID1001	Shift materials using manual handling methods
	TLIE1003	Participate in basic workplace communication
	TLIF1001	Follow work health and safety procedures
Level 1	TLIF2010	Apply Fatigue Management Strategies
Flashbutt Welding	TLIF2080	Safely Access the Rail Corridor
	TLIU2008	Apply Environmental procedures to rail infrastructure
	CPCCOHS1001A	Work Safely in the Construction Industry (White Card)
	TLIF1002	Conduct housekeeping activities
	TLIW2012	Grind Rails
		Plus 3 electives from Electives level 1 and 2

Level 2 Flashbutt Welding	Units	Unit Description
	TLIB2092	Operate Minor Mechanical Equipment
	TLIF2081	Perform Lookout duties

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **48** of **53** 

TLIF3083	Conduct track protection assessment
RIIWHS204D	Work safely at heights
RGWA01	GWA Rules and Addendum
TLIK2010	Use infotechnology devices in the workplace
TLIJ2001	Apply quality procedures
TLID3043	Overhead Crane
TLIB1093	Clean equipment and restore worksite
	Plus 3 electives from Electives level 1 and 2

	Units	Unit Description
	RIICCM201D	Carry out measurements and calculations
	TLIE2007	Use Communication systems
Level 3 Flashbutt Welding	TLIF3003	Implement and monitor work health and safety procedures
	TLIJ3002	Apply quality systems
	TLIB2001	Check and assess operational capabilities of equipment
	RIICM203D	Read and Interpret plans and job specifications
	Pl	us 3 electives from Electives level 1 and 2 or 3 and 4

	Units	Unit Description
	TLIE2029	Conduct Workplace Information Briefings
Level 4	TLIE2008	Process workplace documentation
Flashbutt Welding	TLIS3010	Test rail using ultrasonic equipment
	TLIF2006	Apply accident-emergency procedures
	TLIS3011	Test rail using non-destructive testing equipment
	Plu	s 5 electives from Electives level 1 and 2 or 3 and 4

#### Electives Levels 1&2

Units	Unit Description
HLTAID003	Provide first aid
TLIF2082	Perform Handsignaller duties (Handsignalling Level 1)
TLIF2082	Perform Handsignaller duties (Handsignalling Level 2)
TLILIC2016	Licence to drive a heavy rigid vehicle
TLILIC3017	Licence to drive a heavy combination vehicle
TLIA3039	Receive and store stock
TLIS2013	Install minor structures
TLIS2027	Install and maintain surface track drainage
TLIS2028	Install and replace transoms
TLIB2097	Install and maintain guard rails
TLIS2035	Install and repair fences and gates
TLIE2001	Present routine workplace information
TLIJ2001	Apply quality procedures
RIICCM205	Carry out manual excavation
RIICCM207	Spread and compact materials manually
RIICCM208	Carry out basic levelling

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **49** of **53** 

CPCCCM2008	Erect and dismantle restricted height scaffolding
TLID1002	Shift a load using manually-operated equipment
RIIWHS205	Control traffic with a stop-slow bat
TLID3033	Operate a vehicle-mounted loading crane
TLID3035	Operate a boom type elevating work platform
RIIHAN301	Operate elevating work platform
TLILIC2015	Licence to drive a medium rigid vehicle
TLIC2025	Operate four wheel drive vehicle
TLID2010	Operate a forklift
NWP227	Control vegetation on a site
TLIB1030	Undertake general site maintenance
RIIMPO319	Conduct backhoe/loader operations
RIIMPO320	Conduct civil construction excavator operations
RIIMPO322	Conduct civil construction tracked front end loader operations
RIMPO321	Conduct civil construction wheeled front end loader operations
RIIMP0318	Conduct civil construction skid steer loader operations
TLIC2059	Propel and operate light on-track equipment
TLIB2121	Maintain rail joints
AHCARB205	Operate and maintain chainsaws
CPCCCO2014	Carry out concrete work
TLIW2001	Operate under tack protection rules
TLIS2031	Install railway sleepers
TLIS2034	Install and repair rail fastening systems
TLIF2062	Apply awareness of safeworking rules and regulations
TLIK2010	Use infotechnology devices in the workplace
CPCCCM2007B	Use explosive power tools
RIIMP0317D	Conduct roller operations
RIIMPO326D	Conduct civil construction water cart operations
RIIWHS302D	Implement traffic management plan
TLIB2034	Maintain poles and associated hardware
TLIC3045	Operate road/rail vehicle
TLIS2020	Install overhead wiring structure
TLIW2001	Operate under track protection rules
TLIW2028	Identify the principles of ballast cleaning operations
TLIW2029	Identify the principles of ballast regulator operations
TLIW2030	Identify the principles of dynamic track stabiliser operations
TLIW2031	Identify the principles of self-propelled rail grinder operations
TLIW2032	Identify the principles of tamping machine operations
TLIW2033	Identify the principles of mechanised track laying operations
TLIW2038	Place and remove temporary speed restriction equipment
TLIG1001	Work effectively with others
TLII1002	Apply customer service skills
TLIW2037	Clip and Secure Points
TLIB2091	Measure and record track geometry

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **50** of **53** 

TLIB2085	Apply Track Fundamentals	
TLIB2092	Operate Minor Mechanical Equipment	
TLIS2030	Carry out track ballasting	
TLIS2044	Carry out rail installation	
TLIF2081	Perform Lookout Duties	
TLIF3083	Conduct track protection Assessment	
	ARTC NTSA Part B - NCOP	
RIIWHS204D	Work safely at heights	
TLIB1093	Clean equipment and restore worksite	
RIIWHS202D	Enter and work in confined spaces	
TLIB2084	Carry out routine maintenance of structures	
TLIB2086	Apply Awareness of structures fundamentals	

#### Electives levels 3&4

Units	Unit Description			
TLIW3015	Weld rail using aluminothermic welding process			
TLIW3016	Weld rail using flashbutt welding process			
TLIW0036	Apply electric welding process to rail			
TLIS2004	Install and maintain rail bonding systems			
TLIB3047A	Repair and adjust mechanical signalling equipment and infrastructure			
TLIL3083	Implement a track work authority and manage rail traffic through worksites			
TLIL3084	Implement a local possession authority			
TLIL4069	Plan and coordinate protection for multiple worksites within limits of a work or track authority			
TLIC2054	Access rail track to run track vehicle within a defined worksite			
TLIC0083	Access rail track to travel track vehicle under a proceed authority			
TLIC0084	Access rail track to travel track vehicles under manual block working conditions			
TLIF3013	Coordinate breakdowns and emergencies			
TLIW3005	Perform direct burial of cables			
TLIS3023	Erect and mount structures and housings for signalling equipment			
TLIB3048	Carry out off-site repair, overhaul and assembly of mechanical signalling equipment			
TLIS2012	Install and service rail lubrication equipment			
TLIS2033	Install and repair temporary track supports			
TLIS3037	Install and repair rail earthworks			
TLIS3025	Implement ballast unloading			
TLIF3063	Administer the implementation of fatigue management strategies			
MEM24003B	Perform basic magnetic particle testing			
MEM24001B	Perform basic penetrant testing			
TLIW2012	Grind rails			
TLIB2083	Maintain bridge bearings			
TLIG3002	Lead a work team or group			
CPCCPD3030B	Apply protective paint coating systems			
TLIB2096	Repair concrete/masonry structures			
TLIB2082	Repair steel structures			

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **51** of **53** 

TLIB2081	Repair timber structures		
TLIB2084	Carry out routine maintenance of structures		
TLIB3102	Adjust rail		
TLIB3099	Examine track infrastructure		
TLIS3026	Implement track maintenance and construction		
TLIW3013	Grind switches and crossings		
TLIE3015	Undertake rigger/dogger and driver communication		
MEM05015D	Weld using manual metal arc welding processes		
TLIF3058	Apply safeworking rules and regulations to rail functions (Safeworking Level 1)		
TLIF3058	Apply safeworking rules and regulations to rail functions (Safeworking Level 2)		
TLIC2058	Travel medium or heavy self-propelled on-track equipment		
TLID3033	Operate a vehicle-mounted loading crane		
TLIB3058	Maintain aerial signal/telecommunications lines and cables		
TLIK2010	Use infotechnology devices in the workplace		
TLIF3003	Implement and monitor occupational health and safety procedures		
TLIJ3002	Apply quality systems		
TLIU4001	Implement and monitor environmental protection policies and procedures		
TLIS2036	Use chemical repair products		
RIICCM201	Carry out measurements and calculations		
TLIW3035	Heat and cut materials using oxy-LPG equipment		
· · · · · ·	Decommission mechanical signalling infrastructure equipment and interlocking		
TLIS4007	equipment from service		
TLIS3011	Test rail using nondestructive testing equipment		
TLIS3010	Test rail using ultrasonic equipment		
TLIC002078	Identify and respond to signals and trackside signs		
TLIS3045	Install Turnouts		
TLIS2044	Carry out rail installation		
TLIL3065	Implement a track occupancy protection		
TLIB2001	Check and assess operational capabilities of equipment		
TLIW3035	Heat and cut materials using oxy/LPG equipment for the rail industry		
AHCARB205A	Operate and maintain chainsaws		
AHCARB202A	Fell small trees		
NWP227B	Control vegetation on a site		
RIIRIS401D	Apply site risk management system		
TLIB4071	Install and maintain pole mounted switches and transformers		
TLIC2081	Pilot traffic within work on track authority limits		
TLIS2027	Install and maintain surface track drainage		
TLIE2001	Present routine workplace information		
TLIE2007	Use Communication Systems		
TLIE2029	Conduct workplace Information Briefings		
RIICM203D	Read and Interpret plans and job specifications		
TLIF2006	Apply accident-emergency procedures		
TLIB3100	Visually Inspect Track Infrastructure		
TLIE2008	Process workplace documentation		
TLIB3095	Check and Repair points and crossings		

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **52** of **53** 

TLIB3094	Check and Repair Track Geometry			
TLIB3002	Test equipment and isolate faults			
TLIS3039	Measure and mark track for resurfacing			
TLIS3039	Implement structures maintenance and installation of minor structures			
TLIB3098	Examine concrete/masonry structures			
MEM18001C	Use hand tools			
MEM24012C	Apply metallurgy principles			
TLIB3046	Service and clean mechanical signalling equipment and infrastructure			
TLIS3005	Install mechanical infrastructure for signalling			
TLIS3009	Install mechanical signalling locking and interlocking devices			
TLIB3047	Repair and adjust mechanical signalling equipment and infrastructure			
TLIB3053	Maintain mechanical signalling locking and interlocking devices			
UETTDRRF02B	Perform pole top rescue			
TLIB3120	Test mechanical signalling equipment and isolate faults			

ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 Page **53** of **53** 

# ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017

Signed for and on behalf of Australian Rall Track Corporation by

Name:	Nathan Farr
Position:	General Manager People
Address:	5/33 Newton Street Broadmeadow, NSW 2292
Signature:	
Date:	5/12/17
Witness:	Merrilyn Beer
Signature:	Res

Signed for and on behalf of the Australian Rall Tram and Bus Industry Union by

Name:	DARREN BRETT PHILLIPS		
Position:	SA /NT BRANCH SECRETARY		
Address:	63 LEDGER ROAD BEVERLEY	SA	5009
Signature:	Cà'A		
Date:	7 December 2017		
Witness:	Kaye Brown	والالاستناسي	
Signature:	<u></u>		·

Signed for and on behalf of the Automotive, Food, Metals, Engineering, Printing and Kindred Industries Union by

Name:		
Position:	************************************	
Address:		
Signature:		
Date:	Access and an an an and a second s	
Witness:	jes	
Signature:	. <u>Na mana kataka kao amin'ny fisiana amin'ny fi</u>	

# ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017

#### Signed for and on behalf of Australian Rail Track Corporation by

Name:	Nathan Farr
Position:	General Manager People
Address:	5/33 Newton Street Broadmeadow, NSW 2292
Signature:	
Date:	5/12/17
Witness:	Merrilyn Beer
Signature:	- i Jeer

Signed for and on behalf of the Australian Rail Tram and Bus Industry Union by

Name:	<u></u>		•	
Position:		 		
Address:	·	 · · · · · · · · · · · · · · · · · · ·		
Signature:		 	1. Jagan (1997) - 1. Jagan (1997)	*
Date:				
Witness:		 . <u></u>		
Signature:	<u> </u>	 		

Signed for and on behalf of the Automotive, Food, Metals, Engineering, Printing and Kindred Industries Union by

Name:	PETER BAUER
Position:	ASSISTANT STATE SECRETARY
Address:	53-61 DALE ST. BRT ADELAIDE SOIS
Signature;	Lalu
Date:	1 DECEMBER 247
Witness:	DEREK WINTER
Signature:	Nevek whinters

#### IN THE FAIR WORK COMMISSION

FWC Matter No.:

AG2017/6055

#### Applicant:

Australian Rail Track Corporation

# **Undertaking-section 190**

I, Nathan Farr, General Manager People, of Australian Rail Track Corporation give the following undertakings with respect to the ARTC South Australia / Western Australia Infrastructure Maintenance Enterprise Agreement 2017 ("the Agreement"):

- 1. I have the authority given to me by Australian Rail Track Corporation to provide this undertaking in relation to this application before the Fair Work Commission.
- 2. In respect of Employees covered by the Agreement, Australian Rail Track Corporation undertakes that:
  - a) Notwithstanding clause 2.7, employees with at least 2 years' but less than 3 years' service are entitled to redundancy in accordance with the National Employment Standards.
  - b) Notwithstanding clause 3.1.2.3, Adult Apprentices will be paid at the rate in the Rail Industry Award if that rate is higher than the rate in clause 3.1.2.3.
  - c) A shiftworker who is a seven day shiftworker, regularly rostered to work on Sundays and public holidays, will receive an additional weeks' annual leave.
  - d) Notwithstanding Clause 4.1.3, an employee can accrue up to 8 weeks (10 weeks for shiftworkers) annual leave before an employee has an excessive leave accrual which can result in a direction to take leave with notice of not less than 8 weeks or more than 12 months.
  - e) Where an employee's employment is terminated in accordance with clause 4.2.5.2 the employee will receive the applicable notice or payment in lieu.
  - f) Notwithstanding clause 4.5.2, the taking of Community Service Leave in relation to an emergency service is not dependent on whether the employee is required to attend to Australian Rail Track Corporation assets.

Employee name: Nathan Farr

Signature:

Date: 10 April 2018